4063 5 報告番号 日 第

Micromanipulation System for Handling of Biological Molecule and High Throughput Screening of Microbes

Keisuke Morishima

Micromanipulation System for Handling of Biological Molecule and High Throughput Screening of Microbes

Keisuke Morishima

Department of Micro System Engineering Nagoya University



Acknowledgement

It is my pleasure to acknowledge many people who have contributed their time to helping me with this work. First, I would like to extend my appreciation and heartfelt gratitude to Prof. Toshio Fukuda, Department of Micro System Engineering, Nagoya University (presently, Professor of Center for Cooperative Research in Advanced Science and Technology, Nagoya University), for his guidance, advice, and encouragement throughout the course of this work.

I would like to express my gratitude to Prof. Koji Ikuťa, Department of Micro System Engineering, Nagoya University, and Prof. Kazuo Sato, Department of Micro System Engineering, Nagoya University, for having served on my doctoral committee.

I am particularly grateful to Prof. Fumihito Arai, Mr. Hideo Matsuura, Dr. Hidenori Ishihara, and Dr. Kosuke Sekiyama, Department of Micro System Engineering, Nagoya University, for their guidance, and a number of stimulating discussions.

I gratefully acknowledge Prof. Kenichi Yoshikawa, Graduate School of Human Informatics, Nagoya University, Prof. Masao Washizu, Department of Mechanical engineering, Kyoto University, Prof. Nobuyuki Uozumi, Bioscience Center, Nagoya University, and Prof. Yoshisnobu Baba, Department of Medicinal Chemistry, Faculty of Pharmaceutical Sciences, The University of Tokushima for their technical advice and fruitful discussion. I also deeply acknowledge Prof. G. R. Fuhr, Lehrstuhl für Membranphysiologie, Institut für Biologie, Humboldt-Universität zu Berlin, for valuable discussion during his stay in my laboratory. I am most grateful to Dr. M. Ueda, The Institute of Scientific and Industrial Research, Osaka University and Mr. H. Oana, Department of Applied Physics, Nagoya University for their technical and experimental advice and fruitful discussion. I am particularly grateful to Mr. Y. Yamasaki, Mr. S. Nomura and all the members of Prof. Yoshikawa Laboratory for their technical and experimental advice and fruitful discussion. I wish to acknowledge Mr. Koji Horio, who worked at Moritex Co., Ltd. for technical assistance and valuable discussion. I also acknowledge Mr. Y. Ogino, all the members of Prof. Fukuda Laboratory, who cooperated to complete this thesis and made my student life happier. I also thank all secretaries of Prof. Fukuda Lab. for their supports in my miscellaneous work. Furthermore, I thank my friends for my meaningful life in the university.

At last but not at least, I owe my lasting gratitude to my parents, Yasuo Morishima and Hiroko Morishima, who enabled me to continue my undergraduate and graduate studies.

This work was partly supported by Research Fellowships of the Japan Society for the Promotion of Science for Young Scientists (JSPS Research Fellowships for Young Scientists) and the Sasakawa Scientific Research Grant from The Japan Science Society.

> January 1998 Keisuke Morishima

Contents

1	Inti	roduction :			
	1.1	Micromanipulation			
		1.1.1	.1 Micromanipulator		
		1.1.2	Needs for Bio-Micromanipulation	2	
	1.2	Biome	edical Application to Micromanipulation in MEMS \ldots .	8	
		1.2.1	Micro Fluidic Devices and Micro Chip Technology \ldots	8	
		1.2.2	Human Genome Project and DNA Sequencing	11	
		1.2.3	High Throughput Screening of Microbes	15	
	1.3	Motiv	ation and Goals	16	
	1.4	Organ	ization of this Dissertation	17	
2	Nor	nconta	ct Transportation of DNA Molecule by Dielectrophor	etic	
	For	ce		20	
	2.1	Introd	uction	20	
	2.2	Manip	oulation of DNA Molecules	23	
		2.2.1	Micro DNA Flow System	23	
		2.2.2	Gel electrophoresis	23	
		2.2.3	Manipulation of DNA molecules by Dielectophoretic Force	24	
		2.2.4	Concept and principle of selective transportation system		
			of DNA molecules	25	
	2.3	Migrat	tion Model of DNA Molecule between Micro Electrodes .	27	
		2.3.1	Structure	27	
		2.3.2	Model of electric field \ldots \ldots \ldots \ldots \ldots \ldots	29	
		2.3.3	Calculation of Dielectrophoretic Force and model of mo-		
			tion of the DNA molecule	32	

	2.4	Exper	iments	34
		2.4.1	Experimental apparatus	34
		2.4.2	Materials	35
		2.4.3	Experimental results	35
	2.5	Discus	sion	36
	2.6	Summ	ary	41
3	Tra	nsport	ation of DNA Molecule by Dielectrophoretic Force	
	Uti	lizing (the Conformational Transition in the Higher Order	
	Stru	ucture	of DNA	45
	3.1	Introd	uction	45
	3.2	Confo	rmational Transition of DNA and Transportation of DNA	
		Molec	ule by dielectrophoresis	47
		3.2.1	Dielectrophoretic Force acting on the DNA Molecule in	
			a Nonuniform Field	47
	3.3	Confo	rmational Transition in the Higher Order Structure of DNA	48
	3.4	Transp	portation of DNA Molecules in the Globule State	50
	3.5	Exper	iments \ldots	51
		3.5.1	Materials	51
		3.5.2	Experimental apparatus 	52
		3.5.3	Experimental condition	53
		3.5.4	Results	54
	3.6	Discus	sion	56
	3.7	Summ	ary	57
4	Scre	eening	of a Single Escherichia coli in Microchannel by Elec-	
	tric	Field	and Laser tweezers	66
	4.1	Introd	uction	66
	4.2	High 7	Throughput Screening of Escherichia coli	67
		4.2.1	Operation Improvement by Dielectrophoretic Force and	
			Laser tweezers	67
		4.2.2	Dielectrophoretic dispersion of Escherichia coli	68
		4.2.3	Optical Trapping of Escherichia coli by Laser Tweezers	69

	4.3	$\mathbf{Experiments}$	70
	4.4	Results and Discussion	72
	4.5	Summary	74
5	Bio	-Micromanipulation System for High Throughput Screen-	
	\mathbf{ing}	and Micro Surgery in Microchannel	83
	5.1	High Throughput Screening System of Microbes	83
	5.2	Important Technologies for Bio Micro Manipulation	84
	5.3	New methodology of selective transportation system of Escherichia	
		coli	87
	5.4	Application of Liposome to Micro Encapsulation Delivery of	
		Escherichia coli	90
		5.4.1 Giant Liposome as Micro Capsule	90
		5.4.2 Optical Trapping of Liposome by Laser tweezers	90
	5.5	Summary	91
6	Cor	nclusion	94
	6.1	Summary	94
	6.2	Future work	96

.

List of Figures

1.1	Biological Application Field of Micromanipulation	2
1.2	Micro/Nano World	4
1.3	Micromanipulation with MicroGripper [7]	5
1.4	Biosynthesis based on Field Control	7
1.5	History of the Development of the Biotechnology and Applica-	
	tion to Micromanipulation in MEMS	9
1.6	History of the Development of the Biotechnology and DNA	
	Technology	12
1.7	Genetic Analysis	13
1.8	Weakpoint of Gel Electrophoresis	14
1.9	Organization of this Dissertation	19
2.1	Polarization of DNA molecule and dielectrophoretic force	26
2.2	Micro DNA Flow System	28
2.3	Line packet	28
2.4	Circle packet	29
2.5	Electrode design in an experiment	30
2.6	The model of the coordinate system for electric field calculation	31
2.7	Schematic diagram of experimental apparatus	35
2.8	The Variation of the Velocity of the Migration of DNA Molecule	
	with the Electrode Gap for the angle $\delta=30^\circ$ (square) between	
	the electrodes and the angle $\delta = 60^\circ$ (circle) between the elec-	
	trodes with best fits of Eq. (2.20) (lines) \ldots \ldots \ldots	37
2.9	Fluorescence Images of T4DNA Molecule Suspending in a Medium	
	(at E=0)	38

2.10	Transportation Trajectory that the DNA Molecule Migrates	
	Straight Between the Electrodes	39
2.11	The Variation of the Electric Field Strength with the Electrode	
	Gap for the angle $\delta = 30^\circ$ (square) between the electrodes and	
	the angle $\delta = 60^\circ$ (circle) between the electrodes	42
2.12	The Variation of the Field Gradient gradE2 with the Electrode	
	Gap for the angle $\delta = 30^\circ$ (square) between the electrodes and	
	the angle $\delta = 60^{\circ}$ (circle) between the electrodes	43
3.1	Concept of Micro DNA Flow System	47
3.2	Polarization of Globular DNA Molecule and dielectrophoretic	
	Force	48
3.3	Conformational change of DNA molecule	49
3.4	$Fluorescent\ image\ of\ conformational\ transition\ of\ DNA\ molecule$	
	[35]	50
3.5	Schematic diagram of experimental apparatus	52
3.6	Schematic diagram of the observation sample and the experi-	
	mental system	53
3.7	Design of Packet in an experiment	54
3.8	Compacted globule DNA molecule suspending in a medium (at	
	E=0)	58
3.9	Transportation trajectory that the DNA molecule migrates straight	
	between the electrodes near X1 point (as shown in Fig. 3.7) \therefore	59
3.10	Transportation trajectory that the DNA molecule turns to the	
	left near X2 point (as shown in Fig. 3.7)	60
3.11	Transportation trajectory that the DNA molecule turns to the	
	right near X3 point (as shown in Fig. 3.7)	61
3.12	Transportation trajectory that the DNA molecule near X4 point	
	(as shown in Fig. 3.7) migrates to the left toward the $\mathrm{Ch2}$	62
3.13	Transportation trajectory that the DNA molecule near X4 point	
	(as shown in Fig. 3.7) migrates to the right toward the $\mathrm{Ch3}$	63

3.14	Transportation trajectory that the DNA molecule migrates in	
	a zigzag through X4 point (as shown in Fig. 3.7) by switching	
	the electric field (X: switching point)	64
3.15	Calculation of transportation trajectory of DNA molecule around	
	the micropacket $(\delta 1, \delta 2) = (120^{\circ}, 30^{\circ})$ from X1 point to X4 point	65
3.16	Calculation of transportation trajectory that the DNA molecule	
	near X4 point migrates to the left toward the Ch2 in the mi-	
	cropacket $(\delta 1, \delta 2) = (120^\circ, 30^\circ)$	65
3.17	Calculation of transportation trajectory that the DNA molecule	
	near X4 point migrates to the right toward the Ch3 in the mi-	
	cropacket $(\delta 1, \delta 2) = (120^\circ, 30^\circ)$	65
4.1	Electrical dispersion of Escherichia coli by dielectrophoretic force	69
4.2	Schematic diagram of the observation sample $\ldots \ldots \ldots$	70
4.3	Schematic diagram of the electrode design $\ldots \ldots \ldots \ldots$	71
4.4	Schematic diagram of experimental apparatus	72
4.5	Microscope image of Escherichia coli trapped by Laser tweezers	
	in an experiment, while the high frequency field switches off (
	$E{=}0$)	75
4.6	Sequence picture 1 that the Escherichia coli in the state of pearl	
	chain formation between the electrode2 and the electrode3 mi-	
	grate toward the Ch2, while the high frequency field switches	
	on	76
4.7	Sequence picture 2 that the Escherichia coli in the state of pearl	
	chain formation between the electrode2 and the electrode3 mi-	
	grating toward the Ch2, while the high frequency field switches	
	on	77
4.8	Microscope image of Escherichia coli trapped by Laser tweezers	
	in an experiment, while the high frequency field switches off (
	E=0)	78
4.9	Transportation of the Escherichia coli in the state of pearl chain	
	formation between the electrode2 and the electrode3 migrating	
	toward the Ch2	79

.

ę

4.10	Aggregation state of the Escherichia coli while the high fre-	
	quency field switches off between the parallel electrodes \ldots .	80
4.11	Aggregation state of the Escherichia coli in the state of pearl	
	chain formation, while the high frequency field switches on be-	
	tween the parallel electrodes $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	81
4.12	Aggregation state of the Escherichia colis in the state of pearl	
	chain formation near the entrance of channel 2 while the high	
	frequency field switches on	82
5.1	Micro Glass Chip for High Throughput Screening of Microbes .	87
5.2	System flow diagram for high throughput screening of Escherichia	
	coli	88
5.3	Schematic diagram of Fluorescence Microscopy with Laser Ma-	
	nipulator	89
5.4	Schematic diagram of Liposome	91
5.5	Concept of trapping Single globular DNA by Laser tweezers and	
	its Application to Micro encapsulation by Liposome	92
5.6	Microscopic Image of Liposome trapped by Laser tweezers	93

•

•

.

.

List of Tables

1.1	Methods of Controlling the Energy Field in Bio Micromanipu-	
	lation	8
1.2	Comparison of the Energy Field	8
2.1	The results of calculation of parameter k_d	40

Chapter 1 Introduction

1.1 Micromanipulation

1.1.1 Micromanipulator

Many types of robotic manipulators have been studied, some of which are aiming at handling small objects ranging from millimeter to micrometer order of size. Grasping and clamping are required to handle much smaller objects. Fukuda et. have reported a bilateral control method [1] of such micromanipulators, requiring the contractive position and force controls, and also a visual recognition method [2] of objects for the micromanipulator. It is advantageous to develop a small manipulator itself in size for handling small objects rather than to use the conventional relatively larger size of manipulator. Thus, the actuator employed at such a micromanipulator is required to be much smaller than ever and to have more degrees of freedom. In this sense, it turns out to be difficult to use an ordinary electro-magnetic motors. Therefore, the development of a new actuator [3–5] has been desired.

1.1.2 Needs for Bio-Micromanipulation

Recently, bio-science is progressed so much with the advancement of the bio-technology, such as gene engineering, cell engineering, as shown in Fig.1.1.



Figure 1.1: Biological Application Field of Micromanipulation

In these research fields, operation of the micro/nano manipulation, mass production, repetitive processing, and high speed and high precision processing are required. For the breakthrough in these fields, integration of the distributed research fields and system technologies is important. At this moment, conventional robotics has potentiality for the breakthrough in the bio-engineering. We should integrate the robotics and bio-engineering. We call this field of study as the bio-robotics [8].

Basic technologies of the bio-robotics are manipulation technology, micro system technology, visualization technology, human interface technology to improve operability, automation technology, and so on. Especially, micro/nano manipulation of a DNA molecule, animal or plant cell, and embryo is important. However, research works are not enough at present.

Micromanipulation can roughly be classified into contact type [6, 7] and noncontact type. We present a new direction and new methods of the noncontacttype bio micromanipulation.

As an example of the contact tasks with the conventional micromanipulator, nuclear transplantation or embryo culture requires those technologies. Most of the bio micromanipulation tasks are performed in water solution. Thus, compared with the general manipulation tasks in the air [6,7], we have to consider many different kinds of forces acting on the cell in solution such as the gravity, buoyancy, resistance force from viscosity of the fluid, Brownian motion, interactive forces in the micro/nano world, as shown in Fig.1.2, such as Van der Waals force, electrostatic force depending on the surface charges. In the contact manipulation, as shown in Fig.1.3, the internal force is generated at the object. It is important to control the internal force precisely. It is not easy to control its force autonomously. To realize the flexible manipulation, we have to consider an integrated micro force sensor on the microgripper.



Micro/Nano World

Figure 1.2: Micro/Nano World



Figure 1.3: Micromanipulation with MicroGripper [7]

Moreover, since size of each manipulation object is in micron order, it is quite difficult to recognize its configuration and to manipulate it freely. Automation of manipulation is also quite difficult and skills of human operator are required in contact tasks. Moreover, if there is a flow in a solution, fluid force will disturb its motion. To reduce the influence on the environmental condition and control their migration, we have to consider more suitable forces acting on them. And also if we operate the conventional micromanipulator in observing biological cells with a microscope and manipulate them with it, it is difficult to handle them sequentially. In this way, as described above, we have difficulties in developing a micromanipulation system in a contact type, which is more complicated systems and restricted to limited tasks, to realize autonomous function.

On the other hand, if we apply the external force to the biological object in a solution in noncotact ways, not in a cotact way, we can achieve the sequential tasks, such as transportation, manipulation, separation of biological molecules, by controlling its suitable force acting on them. And also, because almost all of the biological materials are soft and flexible materials, it is easier to manipulate them in noncotact ways than in contact ways when we handle biological objects noninvasively.

Then, we consider how to handle a biological molecule in a noncontact way and discuss the methods of controlling the energy field in bio micromanipulation. The methods to apply external force acting on the biological object in a noncontact way are described as follows.

(1) Electric Field

Coulomb Force (D.C. field)

Dielectrophoretic Force (A.C. field) [14, 37, 111, 112]

- (2) Laser Beam [46, 47, 77, 78]
- (3) Ultrasonic Field [39, 40]
- (4) Fluid Flow of Solution

Fig.1.4 depicts the concept of biosynthesis based on field control. Tab.1.1 shows the methods and characteristics of controlling the energy Field in bio



Figure 1.4: Biosynthesis based on Field Control

micromanipulation. Among these for us, Laser Beam and Ultrasonic Field require position information of the object, and the sensing system is needed for automation. When we consider precise and noninvasive handling of micro objects, Laser Beam is suitable tool. Ultrasonic Field will disturb the environment with the fluid flow. Tab.1.2 shows the comparison of the energy field to control the biological object. For these reasons, we can simplify the system by using the electric field and laser tweezers. In case of using coulomb force, convection caused by electroosmosis, polarization of the microelectrode, and electrolysis will occur easily. So, we propose to use dielectrophoretic force to control the biological object, especially giant DNA molecules, and microbes.

	Application	Control Method	Selectivity	
Electric	Transportation	High Frequency Electric Field	Difference of Dielectric	
Field	Separation	Patterning of Microelectrode	Properties of Particles	
Optical Field	Trapping of Single Object	3D Confocal Scaning of Laser Beam with Mirror	Trapping Particles larger than the wave length of Laser Beam	
Ultrasonic	Filtration	Configuration of	Difference of Size and	
Field	Fixation	Micro Resonator	Density of Particles	

Table 1.1: Methods of Controlling the Energy Field in Bio Micromanipulation

Table 1.2: Comparison of the Energy Field

	Transportation	Selectivity	Controllability	Fixation	Resolution	3D Operability
Electric Field	0		O	0	0	O
Optical Field	Δ	Δ	O	O	O	0
Ultrasonic Field	Δ	0	0	O		Δ

 \bigcirc : Good, \bigcirc : Fair, \triangle : Poor

1.2 Biomedical Application to Micromanipulation in MEMS

1.2.1 Micro Fluidic Devices and Micro Chip Technology

Basic technologies required for the bio micromanipulation are classified as follows: separation, transportation, and position orientation, control technology, contact, noncontact manipulation with high speed and high precision. Fig.1.5 shows the history of the development of the biotechnology and biomedical study and application to Micromanipulation in MEMS.

Electrophoresis is a powerful separation technique that has been used for years, especially in biochemistry and biology, for the separation of large charged



Figure 1.5: History of the Development of the Biotechnology and Application to Micromanipulation in MEMS

molecules such as proteins. It is based on the differential migration of charged species under the influence of an electrical field. Although this concept can be implemented in various ways, it has been mainly used as gel electrophoresis in which separations are performed in slabs or tubes of gel.

In 1981, Jorgenson performed the electrophoresis experiment in a small capillary with an internal diameter of $100\mu m$ or less, extending the concept of electrophoresis and achieving enormous plate counts [9]. At high field strengths, the small dimensions of the capillary provide efficient heat dissipation, prevent convection, and separate with theoretical plates in the millions. This extraordinarily high efficiency has been applied to categories of molecules that previously had been separated by HPLC, such as organic oligomers and small organic and inorganic compounds.

Recently, micromachining, microlithography, optical fibers, ultramicroelectrodes, and nanotechnology are having a dramatic impact on analytical chemistry. A. Manz proposed "micro total analysis system" [53, 54, 57, 58]. Microchips with micrichannels and reservoirs provide a way to manipulate small volumes for sample handling or separations [48,50,55,59]. Microfabrication and microfluidic techniques [49,52] have been progressed toward small volume handling and integrated microdevices [51,56] capable of chemical and biochemical analysis. The trend in chip technology [60] has appeared to be toward smaller devices. The novel application of microtechnologies and miniaturisation to the pharmaceutical industry provides tools and techniques for all areas of DNA separation [61–63] and the drug discovery process and clinical diagnostics. The study of system integration [64–67] will be much more needed in future.

Compared with capillary electrophoresis by D.C. electric field, the principle of dielectrophoresis, proposed by H. A. Pohl [37], is defined as the lateral motion imparted on uncharged particles as a result of polarization induced by nonuniform electric fields. Many theoretical and experimental studies have been investigated. Microelectrode structures [18, 22], capable of exerting dielectrophoretic force [14, 16, 17, 19, 99, 103, 104], electrorotational force [10–12,109,118] and travelling field force [15,20,23,101,102,105,106] on bioparticles, are being developed to perform a wide range of applications to bioprocessing functions [21, 26], such as manipulation, trapping, transportation, separation, cultivation [11, 13, 24, 25, 100, 107, 108, 117, 119] of biological molecules.

1.2.2 Human Genome Project and DNA Sequencing

For the Human Genome Project [28], the key to sequencing thousands of unidentified human genes is to automate and speed up the special equipment, such as microfluidics that prepares and sequences DNA samples. Determining the human genome sequence and finding the genes is really just a first step. Fig.1.6 depicts the history the development of DNA Technology. After sequencing, we still need to determine what proteins the genes produce, and what those proteins do in the cell. Once the genetic code for a disease is broken, gene and drug therapies can follow. We still need to know the structure and function of the protein produced by the gene, and how that protein interacts in the environment of the cell.

Genes and the proteins hold the key to unlocking the mysteries of genetic diseases. Once the genetic code for a disease is understood, researchers can begin developing gene and drug therapies for that particular disease. The ultimate goal of the worldwide Human Genome Project is to find all the genes in the DNA sequence, develop tools for using this information in the study of human biology and medicine, and improve human health. Sequencing involves determining the exact order of the four individual chemical building blocks, or bases, that form DNA. The total DNA in a single human cell has approximately 3 billion pairs of the chemical building blocks adenine, projects to detect biological signatures of collected samples and in bioremediation projects



Figure 1.6: History of the Development of the Biotechnology and DNA Technology

to optimize micro-organism action.

When biological researchers want to sequence a section of DNA (Fig.1.7), they clone fragments of that section and then run four nearly identical reactions on those fragments [27]. In these reactions, the four bases are chemically labeled with four different fluorescent dyes. To sequence a section of DNA, researchers first use special enzymes that act as biological scissors to cut DNA at specific points into smaller fragments. They then clone or make hundreds of identical copies of these fragments. When researchers wish to sequence a fragment, they run four nearly identical reactions using that DNA as a template, in which the four bases are chemically labeled with four different fluorescent dyes [84]. An electric current is then applied to the gel, and, because



Figure 1.7: Genetic Analysis

the DNA itself has a negative charge, the fragments migrate from the top to the bottom of the plate. The DNA fragments move at different rates depending on their size: smaller ones move faster than larger ones. As the fragments migrate past a certain point in the gel, a laser beam scans back and forth across the plate, exciting the dyes on the DNA bases. As the fragments pass the laser, the bases are separated from smallest to largest. The fluorescent signals generated by the laser are detected by photomultiplier tubes, and a computer captures, stores, and processes them [68].

Another way to speed up the process is to increase the electric field. The velocity of the DNA increases proportionally. In the current system, however, just increasing the field leads to other problems (Fig.1.8).



Figure 1.8: Weakpoint of Gel Electrophoresis

A higher electric field increases the power dissipation, which increases the temperature in the sieving media. And when the gel heats up and the DNA samples in it, thermal diffusion causes the fluorescent bands to spread out. The bands run into each other and can no longer be identified as individual and distinct bands. This problem can be significantly reduced by using a very thin gel (about 50 micrometers thick) or other sieving media in place of the polyacrylamide gel now being used. The thinner gel means that the temperature gradient across the width of the gel is smaller, and the thermal diffusion of the DNA fragment bands is less.

With this thin sieving media, the instrument can run with an electrical field three to four times higher than that used on the conventional instrument. Thus, the speed of the run increases by the same factor.

Recent advances in genetic-engineering technologies then made it possible to examine and sequence DNA faster and more efficiently than ever imagined.

For instance, because the DNA fragments are moving at a higher velocity when they come to the laser, the laser has to scan across the plates faster. In addition, given the 96 lanes now and the 384 to come in the future.

Other systems being developed include the laser-induced fluorescent detection system, the fluidic and pumping system for the polymer medium, a temperature control system, and analysis software. These improvements plus the microchannel plates themselves add up to major parts of the high-throughput DNA sequencer. The final production system is still down the road.

1.2.3 High Throughput Screening of Microbes

Many biologists are eager to manipulate and separate a single biological particle for their experiment. With the conventional method, it takes so many times to reculture on plates. This method has such strong probability for pure culture of targeted microbes, such as a Escherichia coli. It will be such a promising method of pure culture of microbe for the biologists to discover a new microbe. It is possible to manipulate microbes, such as a Escherichia coli and a yeast cell, with the Laser manipulator we used. Once a target Escherichia coli is trapped at the focal point of the laser beam, we can easily realize noncontact manipulation automatically.

1.3 Motivation and Goals

In this study, the manipulation of DNA molecules is defined as "transportation" and "separation". We analyze the transportation of the DNA molecule by dielectrophoretic force. Since a high field gradient can be obtained between the electrode gaps based on the electrode pattern, the dielectrophoretic force, which is in proportion to the magnitude of the field gradient, is generated between the electrode gaps and it is possible to transport DNA molecule along the flow channel. It is considered to be possible to separate a DNA molecule when DNA molecules are continuously transported. By using micromachine technology, it is possible to miniaturize the electrodes and the flow channels, so that the basic technique of manipulation of individual DNA molecules will be advanced. We show that transportation of DNA with the state of compacted globule is profitable in the future practical application for the separation of giant DNAs such as human gene.

Then we propose new methodology of selective transportation system of Escherichia coli by dielectrophoretic force and optical radiation pressure and carried out a preliminary experiment. We demonstrated the experiments.

16

Thus, the bio-automation will be greatly improved by the proposed micro manipulation technology in the future.

1.4 Organization of this Dissertation

This Thesis consists of 6 chapters. Fig.1.9 shows the organization of this dissertation.

Chapter 2 describes noncontact transportation of DNA molecule by dielectrophoretic force. A simple micro electrode-flow system is designed and experiments were carried out. Experimental demonstration of DNA transportation using dielectrophoretic force and direct observation of the DNA molecule in a non-uniform electric field was carried out with fluorescence microscopy. Motion of the DNA molecule is modeled in terms of the electric field generated by the electrode design.

Chapter 3 describes a new methodology on noncontact transportation of DNA molecules by dielectrophoretic force together with the Micro DNA Flow System. We utilize the conformational transition in the higher order structure of DNA for transportation. We designed a simple micro electrode-flow system. Experimental demonstration of DNA transportation in the globule state using dielectrophoretic force and direct observation of the DNA molecule in a nonuniform electric field were carried out with fluorescence microscopy. We show that transportation of DNA with the state of compacted globule is profitable in the future practical application for the separation of giant DNAs such as human gene. Chapter 4 describes a new methodology of selective transportation system of Escherichia coli by dielectrophoretic force and optical radiation pressure. We have developed a prototype of Microchannel system for high throughput screening of Escherichia coli. Experimental demonstration of noncontact transportation and manipulation of Escherichia coli by dielectrophoretic force and radiation pressure of laser tweezers were carried out with Laser manipulator system. In experiments, we show that transportation and separation of Escherichia coli with electric field and optical radiation pressure is profitable in the future practical application for the high throughput screening of microbes.

Chapter 5 describes Bio-Micromanipulation system for high throughput screening and micro surgery in microchannel. We discussed research issues in the bio micromanipulation and presented a new direction in this field. We consider the basic strategies to improve the working efficiency and the operability of the bio micromanipulation. Application of liposome to micro encapsulation delivery and concept of trapping single globular DNA by laser tweezers and its application to Micro encapsulation by liposome is shown.

Chapter 6 summarizes the results presented in this dissertation and discuss research directions for future works.

18



Figure 1.9: Organization of this Dissertation

Chapter 2

Noncontact Transportation of DNA Molecule by Dielectrophoretic Force

2.1 Introduction

Recently, there has been great interest in the sequencing of the genome of various species, especially the human genome. The DNA molecule consists of a sequence of units, called nucleotides, which consist of a phosphate and a sugar group with a base attached. There are four types of nucleotide in the DNA. Physically, the DNA is carried in structures called chromosomes. Each human has 46 chromosomes. The total length of an individual's DNA is about $3x10^9$ bases [27]. At present, only 4% of the sequencing of the human genome has been decoded. It is necessary to develop a technique by which DNA molecules can be separated by the length. It would be expected to separate a giant DNA molecule $(Mbp \sim 10Mbp)$ individually [27].

On the other hand, since remarkable progress has been made in the development of micromachining technology, micro devices such as the microelectrode, micropump, and micromotor have been fabricated on Silicon wafer. Micro electro mechanical systems have been considered for application in the field of biotechnology. For example, the manipulation of many kinds of micro organisms such as a cell, a single DNA molecule, a chromosome etc is a very essential technique. In conventional studies, cell manipulation on a Fluid Integrated Circuit (FIC) [112,113,115], and the trapping of biological particles by use of ultrasonic resonance fields [39,40], and optical tweezer [72,76–78] have been presented. Micromachine technology makes it possible to fabricate microelectrodes and microflow channels [48,54] of the same size as biological particles, so that this technology has great potential to separate micro organisms such as DNA molecules.

At present, gel electrophoresis is the basic technique used in the separation of DNA molecules. However, each research group carries out the experiments only to have to search for the best conditions for separating a giant DNA molecule within the limit of separation [27]. In gel electrophoresis, it takes too much time to separate DNA macromolecules and it is very difficult to separate them in gel. As for the point of improving the durability of the electrodes and the controllability of the manipulation of a single DNA molecule, electrophoresis in a DC field causes electrochemical reaction and electroosmosis so often that it is not suitable for manipulation of DNA molecules. To solve these problems, a high frequency AC field is suitable [37, 110, 112, 113, 115]. For this reason, the study of cell manipulation by using the dielectrophoretic force has been reported [14–16, 19, 41, 110, 112, 113, 115]. One of the applications of the dielectrophoretic force to DNA molecules is, for example, the electrostatic orientation of the DNA molecule [113–116]. The application of noncontact manipulation of a single DNA molecule by the dielectrophoretic force has not been reported so far. In our study, we define the manipulation of DNA molecules as "transportation" and "separation". We analyze the transportation of the DNA molecule by dielectrophoretic force. Since a high field gradient can be obtained between the electrode gaps based on the electrode pattern, the dielectrophoretic force, which is in proportion to the magnitude of the field gradient, is generated between the electrode gaps and it is possible to transport DNA molecule along the flow channel. It is considered to be possible to separate a DNA molecule when DNA molecules are continuously transported, especially in the circle packet (as shown in Fig. 2.4) as we proposed. By using micromachine technology, it is possible to miniaturize the electrodes and the flow channels, so that the basic technique of manipulation of individual DNA molecules will be advanced.

This chapter proposes the concept of Micro DNA Flow System, as shown in Fig. 2.2. In our system, DNA molecules can be manipulated and transported and finally be separated individually by using the dielectrophoretic force. At first, we developed a simple electrode design on a glass plate to show the possibility of transporting DNA molecules based on Micro DNA Flow System. In the experiments, we observed the motion of DNA molecules between the electrodes and measured the velocity of migration of a single DNA molecule. Thus, the possibility of transportation of DNA molecules can be shown.

2.2 Manipulation of DNA Molecules

2.2.1 Micro DNA Flow System

In order to manipulate and separate an arbitrary single DNA molecule in a fluorescence microscopy system, it is essential to have the following points in Micro DNA Flow System:

- (1) noninvasiveness
- (2) selectivity
- (3) noncontact
- (4) high sensitivity
- (5) high speed
- (6) durability
- (7) bioautomation process

Although electrophoresis is one of the most useful separation techniques, those points; (2), (4), (5), (6), (7) must be improved.

2.2.2 Gel electrophoresis

At first, we explain the difference between the electrophoretic force and the dielectrophoretic force acting on the DNA molecule.

Gel electrophoresis is a method of distinguishing entities according to their mobility in a DC electric field. The electrophoretic mobility of a molecule is influenced by its size and electric charge, especially in gel [27]. The DNA molecule moves to the anode in a uniform field, because it has minus charge in gel. By gel electrophoresis, however, it is impossible to separate a giant DNA molecule(more than 10 Mbp). In addition, it takes too much time and it costs too much to sequence it.

2.2.3 Manipulation of DNA molecules by Dielectophoretic Force

The dielectrophoretic force acting on the DNA molecule is generated in a nonuniform field [38]. The polarization is generated by an external electric field per unit volume, so in general, the Equation of the dielectrophoretic force F_d is expressed as follows [37],

$$F_d = 2\pi a^3 \epsilon_m R_e \left[\frac{\kappa_p - \kappa_m}{\kappa_p + 2\kappa_m} \right] \bigtriangledown E^2$$
(2.1)

$$j = \sqrt{-1} \tag{2.2}$$

$$\kappa_p = \epsilon_p - j \frac{\sigma_p}{\omega} \tag{2.3}$$

$$\kappa_m = \epsilon_m - j \frac{\sigma_m}{\omega} \tag{2.4}$$

a: Radius of a DNA molecule (m)

E: External electric strength where DNA molecule exists between electrodes (V/m)

 ϵ_p, ϵ_m : Permittivity of DNA molecule and medium

 σ_p, σ_m : Conductivity of DNA molecule and medium

 κ_p, κ_m : Complex permittivity of DNA molecule and medium
ω : Angular frequency.

The polarization is generated by an external electric field per unit volume, so in general, the Equation of the dielectrophoretic force F_d is expressed as follows,

$$F_d = \frac{1}{2}\alpha v gradE^2 \tag{2.5}$$

where α is polarizability (C/m^3) , v is volume (m^3) .

In a high frequency field, the DNA molecule is drawn in the direction of the high field gradient because the dielectrophoretic force operates in the direction of $gradE^2$. The force acting on the DNA molecule in a nonuniform field is shown in Fig. 2.1. The conformation of a single DNA molecule in an aqueous solution is in the coil state. In generating a high frequency AC field, however, the field around the DNA molecule becomes a nonuniform field. So, its conformation transforms. Due to this electric field, the positive and negative charges in the DNA molecule generate toward the electric field by its polarization. As a result, the DNA molecule migrates toward the normal to the line of electric line by dielectrophoretic force.

2.2.4 Concept and principle of selective transportation system of DNA molecules

In the future, noncontact manipulation of a single DNA molecule will become a main technique of gene manipulation. In our study, we propose the selective transportation system of DNA molecules for the purpose of transporting and separating a single DNA molecule by use of the dielectrophoretic force



Figure 2.1: Polarization of DNA molecule and dielectrophoretic force

in Micro DNA Flow System, as shown in Fig. 2.2. Micromachine technology has an advantage in mass production of a various kinds of designs of Micro DNA Flow System.

This chapter describes two designs for Micro DNA Flow System. First, in a line packet, as shown in Fig. 2.3, some packets are located along the flow channel, and a single DNA molecule is separated at the point of the packet by controlling the magnitude and the direction of the dielectrophoretic force. They depend upon the electric field generated by the electrodes lying along the line packet.

In the second design, the DNA molecule is injected from the route A into a circle packet as shown in Fig. 2.4 and in order to have DNA molecules

26

migrated along the circle packet, the electric field between each electrode is generated step by step. Next, at the same time the electric field is turned off, an arbitrary DNA molecule near the packet is drawn into route A or route B. In this way, this circle packet has the advantage of flowing DNA molecules continuously until a target DNA molecule is separated from the aggregation of DNA molecules. And by controlling an electric field between the serial electrodes along it, it can be possible that DNA molecules in aggregation state exhibit conformational change to be be easily separated through the circle packet.

Thus, by using dielectrophoretic force, it is possible to manipulate a single DNA molecule and separate it. In electrophoresis, the total length of the flow channel depends on the electrode gap, but in Micro DNA Flow System it is easier to extend the flow channel.

2.3 Migration Model of DNA Molecule between Micro Electrodes

2.3.1 Structure

We designed a simple electrode design (as shown in Fig.2.5) for the purpose of transporting the DNA molecule in one direction. The electrodes consist of metal tape made by stainless on a slide glass. Its thickness is $30\mu m$. The resin made of polyester was spread on the surface of the electrodes as the insulator. The angles δ between the electrodes are of two kinds; $\delta = 30^{\circ}$, and $\delta = 60^{\circ}$.



Figure 2.2: Micro DNA Flow System



Figure 2.3: Line packet



Figure 2.4: Circle packet

2.3.2 Model of electric field

We calculated a model of electric field in a coordinate system as shown in Fig.2.6. Generally, it is quite difficult to solve Laplace's equation, which expresses potential. For simplicity of calculation, we assume the following conditions:

- (i) Since the thickness of the electrode $(30\mu m)$ is much larger than the size of the DNA molecule(diameter about 2nm), the electrodes are considered to consist of two infinite planes between an interior angle δ .
- (ii) An infinite small gap exists at the origin r = 0.
- (iii) Potentialf is a function denoted by Laplace's equation in a cylindrical coordinate system.



Figure 2.5: Electrode design in an experiment

Here, the absolute coordinates of the DNA molecule (x, y), which exists between the electrodes, is expressed by r, θ .

$$x = r \cos \theta \tag{2.6}$$
$$y = r \sin \theta$$

The field between the electrodes is described by Laplace's equation for the potential $\phi[V]$,

$$\frac{1}{r^2}\frac{\partial^2\phi}{\partial\theta^2} = 0. \tag{2.7}$$

Since $r \neq 0$, we have

$$\frac{1}{r^2}\frac{\partial^2 \phi}{\partial \theta^2} = 0, \theta = 0, \phi = 0; \theta = \delta, \phi = V_0$$
(2.8)

 $\theta = 0, \ \phi = 0; \ \theta = \delta, \ \phi = V_0$ We have

$$\phi = \frac{V_0}{\delta}\theta \tag{2.9}$$



Figure 2.6: The model of the coordinate system for electric field calculation

Since

$$E = -\bigtriangledown \phi \tag{2.10}$$

where \hat{a}_{θ} : unit vector in θ -direction. We have the magnitude of the electric field:

$$|E| = \frac{V_0}{r\delta} \tag{2.11}$$

Hence, the field gradient is given by;

$$|grad|E|^2 = \left(-\frac{2V_0^2}{r^3\delta^2}\right)\hat{a}_r$$
 (2.12)

where \hat{a}_r : unit vector in the *r*-direction. We have the magnitude of the field gradient:

$$|grad|E|^2| = \frac{2V_0^2}{r^3\delta^2}$$
(2.13)

2.3.3 Calculation of Dielectrophoretic Force and model of motion of the DNA molecule

The expression for the dielectrophoretic force F_d is obtained by substituting the Eq. (2.5) into the Eq. (2.13):

$$F_d = -\alpha v \left(\frac{V_0^2}{\delta^2}\right) \frac{1}{r^3} \hat{a}_r \tag{2.14}$$

We have the magnitude of the dielectrophoretic force:

$$|F_d| = \alpha v \left(\frac{V_0^2}{\delta^2}\right) \frac{1}{r^3} \tag{2.15}$$

Next, we derive a dynamic model of a DNA molecule between two electrodes. We assume the following conditions:

- (i) The DNA molecule is a spherical particle.
- (ii) The Coulomb's force acting on the DNA molecule acts from the center of the DNA molecule in the direction of the tangent to the line of electric force.
- (iii) The DNA molecules in the experiment system migrate along the middle direction ($\theta = \delta/2$) between the electrodes, as shown in Fig. 2.5.

Therefore, assume the dynamic model of the DNA molecule is as illustrated in Fig. 2.1. We have the equation for motion as follows:

$$m\frac{d^2r}{dt^2} + 6\pi\eta a\frac{dr}{dt} = F_d(r,t) + F_m(r)$$
 (2.16)

where r is vector of the DNA molecule, m is mass of the DNA molecule, eta is viscosity of the solvent, a is radius of the DNA molecule, $F_d(r,t)$ is the electrostatic force acting on the DNA molecule including dielectrophoretic force except for the Coulomb's force, $F_m(r,t)$ is the external force including gravity except for an electrostatic force.

Let the mass of a single DNA molecule be neglected. The term of the inertial force and the term of the external force in the Eq. (2.16) is negligible. We have as the equation of the motion in the direction of the migration as follows:

$$6\pi\eta a U = F_d(r,t) \tag{2.17}$$

where U is the velocity of migration of the DNA molecules. We can obtain the Eq. (2.18) by substituting the Eq. (2.15) into the Eq. (2.17).

$$6\pi\eta a U = \alpha v \left(\frac{V_0^2}{\delta^2}\right) \frac{1}{r^3}$$
(2.18)

$$r = \frac{d}{2\tan\frac{\delta}{2}} \tag{2.19}$$

(as shown in Fig. 2.5)

We can get the velocity U of migration of the DNA molecule by substituting the Eq. (2.19) into the Eq. (2.18) as follows:

$$U = \frac{\alpha v V_0^2}{6\pi \eta a} \frac{1}{\delta^2 r^3}$$

= $\frac{4\alpha v V_0^2}{3\pi \eta a} \frac{\tan^3 \frac{\delta}{2}}{\delta^2} \frac{1}{d^3}$ (2.20)
= $k_d \frac{\tan^3 \frac{\delta}{2}}{\delta^2} \frac{1}{d^3}$

$$k_d = \frac{4\alpha v V_0^2}{3\pi \eta a} \tag{2.21}$$

where k_d is the physical characteristic parameter of the DNA molecule in the electric field.

2.4 Experiments

2.4.1 Experimental apparatus

The experimental system is schematically shown in Fig.2.7. With the fluorescence microscopy, it is easier to observe the dynamics of DNA molecule in an aqueous solution and gel, and perform a quantative analysis [29–31,83]. Fluorescence images of DNA molecules were observed by use of a Zeiss Axiovert 135 TV microscope equipped with a 100 x oil-immersed objective lens and recorded on videotapes with the sampling rate of 30 frames/s through the high-sensitive Hamamatsu SIT TV camera. The data were analyzed with an image processor, Arugus 10 (Hamamatsu Photonics). The applied voltage was controlled by an arbitrary waveform generator.



Figure 2.7: Schematic diagram of experimental apparatus

2.4.2 Materials

Bacteriophage T4DNA, 166 kilobase pairs (contour length $55\mu m$), was employed from Nippon Gene. Trisborate(TBE; pH7.8) buffer, T4DNA, YOYO, and 2-mercaptoethanol(2-ME; as an antioxidant) were added in the solution. The final concentrations are as follows: TBE, 45mMTris; 45mM boric acid, EDTA, 1mM; T4DNA, 0.2mM (in nucleotide); YOYO, 0.1mM; 2-ME, 4%(v/v).

2.4.3 Experimental results

We carried out experiments as follows: the applied frequency is constant 1MHz and the applied voltage is constant 50V. By using two kinds of electrode samples, we measured the velocity of migration of DNA molecule between the electrode gaps.

In the experiments, we determined the dependencies of the veocity of DNA molecule on the difference in the electrode gap and the angle between electrodes, which is one of parameters of the electrode design, as shown in Fig.2.8. Fig.2.8 shows the following points:

- (i) As the electrode gap becomes smaller, the velocity of migration of DNA molecules is faster.
- (ii) As the angle of the electrodes, where the electrode gap is the same size, becomes larger, the velocity of migration of DNA molecules is faster.Because the angle of the electrodes is bigger, the difference in the field gradient is bigger.

Fluorescence images of a T4DNA molecule in an experiment is shown in Fig.2.9. The center of the white circle in Fig.2.9 is a single T4DNA molecule. Fig.2.10 shows the transportation trajectory that the DNA molecule migrates straight between the electrodes.

2.5 Discussion

Using the least square method, we obtained the parameter k_d from the result of the experiments. Though each parameter, related to the physical characteristics of the DNA molecule, is unknown, the coefficient of the term of the dielectrophoretic force acting on a single DNA molecule can be estimated by calculation. The result of the calculation, based on the model and the Eq. (2.20), is shown in Tab.2.1. The solid lines as shown in Fig.2.8 express the



Figure 2.8: The Variation of the Velocity of the Migration of DNA Molecule with the Electrode Gap for the angle $\delta = 30^{\circ}$ (square) between the electrodes and the angle $\delta = 60^{\circ}$ (circle) between the electrodes with best fits of Eq.(2.20) (lines)



Figure 2.9: Fluorescence Images of T4DNA Molecule Suspending in a Medium (at E=0)



Figure 2.10: Transportation Trajectory that the DNA Molecule Migrates Straight Between the Electrodes

result of the calculation using the least square method. Fig.2.8 shows the variation in the velocity of migration of the DNA molecule with the change in the electrode gap. The dispersion of each parameter k_d between two kinds of the electrode samples in the experiments is appeared. Because the parameter k_d depends only upon the viscosity of the medium and the physical characteristics of the DNA molecule, the values are expected to be equal to each other. The

Table 2.1: The results of calculation of parameter k_d

Angle(deg)	k_d
30	2.145×10^{-3}
60	$2.596 imes10^{-3}$

reason why the parameter determined by the experimental results is different is considered in the following points:

- i) The electric field intensity was not obtained precisely because of a measuring error in the width and the angle of the electrode.
- ii) The difference of shape, volume, and polarizability among individual DNA molecules in the experiments.
- iii) The physical parameters including k_d may depend upon the magnitude of the electric field.
- iv) Measuring error caused by heat convection, ion flow, electrification, vibration etc.

- v) The shape of DNA molecule is like a long chain, so it may be impossible to assume it is a spherical particle.
- vi) If the polarization of the DNA molecule becomes unbalanced, it may be possible that the Coulomb's force acts on the DNA molecule in the direction of its migration.

Those are hypotheses, so it is necessary to consider measuring the physical parameters of the DNA molecule and establish a dynamic model of DNA molecule. From the qualitative viewpoint, in terms of the experimental model, what we discussed in this chapter can be confirmed from the model we proposed.

Fig.2.11 shows the dependencies of the electric strength upon the difference in the electrode gap and the angle between electrodes. Fig. 2.12 shows the dependencies of the field gradient $gradE^2$ upon the difference in the electrode gap and the angle between electrodes. Compared with Fig.2.20 and Fig.2.11, Fig.2.12, the field gradient E^2 has more influence on the velocity of migration of the DNA molecule than the electric field strength. The theoretical curve, as shown in Fig.2.12, was calculated by the Eq.(2.13). From this result, it is considered that the field gradient E^2 is a very important parameter in designing a precise electrode to control the motion of DNA molecules.

2.6 Summary

In summary, we carried out experiments involving noncontact transportation of DNA molecules by using dielectrophoretic force. Compared with the



Figure 2.11: The Variation of the Electric Field Strength with the Electrode Gap for the angle $\delta = 30^{\circ}$ (square) between the electrodes and the angle $\delta = 60^{\circ}$ (circle) between the electrodes.



Figure 2.12: The Variation of the Field Gradient gradE2 with the Electrode Gap for the angle $\delta = 30^{\circ}$ (square) between the electrodes and the angle $\delta = 60^{\circ}$ (circle) between the electrodes.

method employing electrophoresis, electroosmosis and chemical reaction on electrodes doesn't occur by use of an AC field. In a Micro DNA Flow System it is also easier to extend the flow channel. We designed a simple electrode, carried out the experiments and measured the velocity of migration of the DNA molecule. We observed the dynamics of DNA molecules between electrodes in an AC field. We showed the possibility of noncontact transportation of a DNA molecule by using dielectrophoretic force. We are planning to create a new model, to realize the separation of DNA molecules, and to propose a manipulation system taking into consideration the conformational changes in the DNA molecule. Thus, the bio-automation will be greatly improved by the proposed micro manipulation technology in the future.

Chapter 3

Transportation of DNA Molecule by Dielectrophoretic Force Utilizing the Conformational Transition in the Higher Order Structure of DNA

3.1 Introduction

This chapter analyzes the transportation of the DNA molecule in the globule state [33–36] by dielectrophoretic force [125]. It is very difficult to transport DNA molecule in the coil state stably. Since a high field gradient can be obtained between the electrode gaps based on the electrode pattern, the dielectrophoretic force, which is in proportion to the magnitude of the field gradient, is generated between the electrode gaps and it is possible to transport a giant DNA molecule along the flow channel. Micromachine technology makes it possible to fabricate microelectrodes and microflow channels of the same size as biological particles, so that this technology has great potential to separate

micro organisms such as DNA molecules. In conventional studies, cell manipulation on a Fluid Integrated Circuit(FIC) [112, 113, 115], cell manipulation by using the dielectrophoretic force [14–16, 19, 41, 110, 112, 113, 115], the trapping of biological particles by use of ultrasonic resonance fields [39, 40], optical tweezer [72, 76–78], the electrostatic orientation of the DNA molecule [113–116], and optical manipulation of single DNA molecule by binding between the latex bead and the DNA molecule [47] have been presented. The application of noncontact manipulation of a single DNA molecule by the dielectrophoretic force and transportation of DNA molecule utilizing the conformational transition [33–36] in the higher order structure of DNA have not been reported so far.

In our system, DNA molecules can be manipulated and transported and finally be separated individually by using the dielectrophoretic force [125], as shown in Fig. 3.1. At first, we developed a simple electrode design on a glass plate to show the possibility of transporting DNA molecules based on Micro DNA Flow System. In the experiments, we observed the motion of DNA molecules in the globule state between the electrodes. Thus, the possibility of separation of DNA molecules can be shown.



Figure 3.1: Concept of Micro DNA Flow System

3.2 Conformational Transition of DNA and Transportation of DNA Molecule by dielectrophoresis

3.2.1 Dielectrophoretic Force acting on the DNA Molecule in a Nonuniform Field

The dielectrophoretic force acting on the DNA molecule is generated in a nonuniform field. The polarizability of the DNA molecule depends not only upon the dielectric characteristic, but also upon its conductivity. Some cells and DNA molecules etc. have conductivity in an aqueous solution. So, their conductivity as well as the dielectric characteristic is very important. In this case, dielectrophoretic force F_d is given by,

$$F_d = 2\pi a^3 \epsilon_m R_e \left[\frac{\kappa_p - \kappa_m}{\kappa_p + 2\kappa_m} \right] \bigtriangledown E^2$$
(3.1)

In a high frequency field, the DNA molecule is drawn in the direction of the high field gradient because the dielectrophoretic force operates in the direction of $\text{grad}E^2$. The force acting on the DNA molecule in a nonuniform field is shown in Fig. 3.2. As a result, the DNA molecule migrates toward the normal to the line of electric line by dielectrophoretic force [125].



Figure 3.2: Polarization of Globular DNA Molecule and dielectrophoretic Force

3.3 Conformational Transition in the Higher Order Structure of DNA

It has been known that the conformational change of a DNA molecule in an aqueous solution depends on the composition, temperature, and concentration of the solvent and that the condensation of a single DNA molecule is induced by polycation, hydrophile polymer, and surfactant under a suitable condition [33]. Generally, the conformation of a polymer chain such as DNA molecule is different from in a good solvent and a poor solvent. The conformation of DNA molecule in a good solvent is in the coil state, and the compactization of a DNA molecule, which is in the globule state, is induced in a poor solvent. Fig. 3.3 illustrates the conformational change of a single DNA molecule and Fig. 3.4 shows fluorescent image of conformational transition of DNA molecule [35,36]. Recently the coil-globule transition has been investigated both theoretically and experimentally [32,34] so it has been possible to produce a DNA molecule in the compact globule state in vitro [36]. It is easy to transform the conformation of a DNA molecule for its transportation since this conformational transition in the higher order structure of DNA is reversible.



Figure 3.3: Conformational change of DNA molecule



Figure 3.4: Fluorescent image of conformational transition of DNA molecule [35]

3.4 Transportation of DNA Molecules in the Globule State

In the future, noncontact manipulation of a single DNA molecule will become a main technique of gene manipulation. In our study, we propose the selective transportation system of DNA molecules utilizing the conformational transition in the higher order structure of DNA for the purpose of transporting and separating a single DNA molecule by use of the dielectrophoretic force in Micro DNA Flow System[3]. It is easier to manipulate a single DNA molecule if the Brownian motion doe'snt have an influence on the structure of DNA molecule. A DNA molecule in the coil state may be torn off during an observation and the transportation since the DNA molecule in the coil state tends to have a great influence of the Brownian motion and the heat convection. So, it is considered that the DNA molecule in the coil state is less suitable for the transportation of DNA molecules. On the other hand, a single DNA molecule both in the globule state and in the coil state can not be manipulated and transported by Laser trapping [47, 76–78], since it is too small.

In this chapter, we demonstrate the transportation of DNA molecule in the globule state by dielectrophoretic force. A DNA molecule is polyelectrolyte [95–98]. However, the compactization of a single DNA molecule is considered to induce charge neutralization, so it is not effective to use the electrophoresis that is one of the methods of separation of charged particles. Dielectrophoresis is suitable for transportation and separation of dielectric particles. In the experiments, we observed the transportation of the DNA molecule in the globule state in 2-propanol.

3.5 Experiments

3.5.1 Materials

Bacteriophage T4DNA, 166 kilobase pairs (contour length $55\mu m$), was purchased from Nippon Gene. Trisborate(TBE; pH7.8) buffer, T4DNA, YOYO, and 2-mercaptoethanol(2-ME; as an antioxidant) were added in the solution. 2-propanol was added in order to keep DNA molecules globule state. The final concentrations are as follows: TBE, 45mMTris; $45mM_{..t}$ boric acid, EDTA, 1mM; T4DNA, 30nM (in nucleotide); YOYO, 6nM; 2-propanol, 40%(v/v); 2-ME, 4%(v/v); saccharose, 0.4g/ml; glucose, 2.3mg/ml; glucoseoxidase, 0.1mg/ml; catalase, 0.018mg/ml.

3.5.2 Experimental apparatus

The experimental apparatus is schematically shown in Fig. 3.5 and Fig.3.6. The observation of DNA molecule with fluorescence microscopy has been carried out [30,31]. With the fluorescence microscopy, it is easier to observe the dynamics of DNA molecule in an aqueous solution and gel, and perform a quantative analysis [30,31]. Fluorescence images of DNA molecules were observed by use of a Zeiss Axiovert 135 TV microscope equipped with a 40 x objective lens and recorded on videotapes with the sampling rate of 30 frames/s through the high-sensitive Hamamatsu SIT TV camera. The data were analyzed with an image processor, Arugus 10 (Hamamatsu Photonics). The applied voltage was controlled by an arbitrary waveform generator.



Figure 3.5: Schematic diagram of experimental apparatus



Figure 3.6: Schematic diagram of the observation sample and the experimental system

3.5.3 Experimental condition

We carried out experiments as follows: the applied frequency is constant 1MHz and the applied voltage is constant 100Vp - p. By using three kinds of electrode samples, as shown in Fig. 3.7, we observed the motion of DNA molecule in the globule state through the packet, as shown in Fig. 3.7, by switching the electric field. We designed three kinds of electrode designs (as shown in Fig. 3.7) for the purpose of transporting the DNA molecule in the globule state. The angles ($\delta 1$, $\delta 2$) between the electrodes are of three kinds; ($\delta 1$, $\delta 2$)=(60° , 60°), (30° , 120°), (120° , 30°). The minimum electrode gap is $100\mu m$.



Figure 3.7: Design of Packet in an experiment

3.5.4 Results

In our previous experiments, we have confirmed the dependencies of the velocity of DNA molecule on the difference in the angle between electrodes, which is one of parameters of the electrode design. As the angle of the electrodes, where the electrode gap is the same size, becomes larger, the velocity of migration of DNA molecules is faster. Because the angle of the electrodes is bigger, the difference in the field gradient is bigger.

From this results, we carried out the experiments by using three kinds of designs of packet and determined the suitable electrode design.

It is necessary to pass between the electrodes in order to transport DNA molecules from Ch1 to Ch2 or Ch3, as shown in Fig.3.7. Using the packet which

has an angle ($\delta 1$, $\delta 2 = (30^{\circ}, 120^{\circ})$ between the electrodes, we observe that DNA molecules migrate from X4 direction to X1 direction. Using the packet which has an angle ($\delta 1$, $\delta 2 = (60^{\circ}, 60^{\circ})$ between the electrodes, it is observed that they aggregate in the middle of the packet between the electrodes. In the case of the packet which has an angle ($\delta 1$, $\delta 2 = (120^{\circ}, 30^{\circ})$, DNA molecules migrate from X1 direction to X4 direction, so they are transported from Ch1 to Ch2 or Ch3. In each cases, DNA molecules don't migrate when the electric field is off. These results are in good agreement with our previous results that the dielectrophoretic force acting on the DNA molecule depends upon the field gradient E^2 . From this reason, we carried out the experiments and observed the motion of the DNA molecules, using the packet which has an angle ($\delta 1$, $\delta 2 = (120^{\circ}, 30^{\circ})$.

Fig. 3.8~Fig. 3.14 shows the transportation trajectory that the DNA molecule migrates between the packet, as shown in Fig. 3.7. Fig.3.8 illustrates that single DNA molecule is in the globule state when the high frequency field is off. It shows that there is no damage in the structure of DNA molecules in the globule state after they are transported between the packet. It is more stable to transport DNA molecule in the globule state than in the coil state and it is suitable for the transportation by dielectrophoretic force.

Fig. 3.10~Fig.3.14 shows the motion of DNA molecules while the high frequency field switches on and off near the packet. They show that by controlling an electric field between electrodes, it is possible that DNA molecules in aggregation state are easily separated through the packet.

55

Fig.3.15~Fig.3.17 show the results of the electric field calculation around the microelectrode configuration. The arrow in Fig.3.15~Fig.3.17 shows the migrating direction of DNA molecule by dielectrophoretic force. In each cases, the observation results of the transportation trajectory of DNA, which is depicted from Fig.3.9 to Fig.3.14, show good agreement with the electric field calculation.

3.6 Discussion

By using dielectrophoretic force, it is possible to manipulate a single DNA molecule and separate it. In electrophoresis, the total length of the flow channel depends on the electrode gap, but in Micro DNA Flow System it is easier to extend the flow channel. And by controlling an electric field between the serial electrodes along it, it can be possible that DNA molecules in aggregation state exhibit conformational change to be be easily separated through the circle packet. Theoretically, we can get the velocity U of migration of the DNA molecule as follows,

$$U = \frac{2}{3\eta} a^{2} \epsilon_{m} R_{e} \left[\frac{\kappa_{p} - \kappa_{m}}{\kappa_{p} + 2\kappa_{m}} \right] \left(\frac{V_{0}^{2}}{\delta^{2}} \right) \frac{1}{r^{3}}$$
$$= \frac{16V_{0}^{2}}{3\eta} \left(\frac{\tan^{3} \frac{\delta}{2}}{\delta^{2} d^{3}} \right) \epsilon_{m} R_{e} \left[\frac{\kappa_{p} - \kappa_{m}}{\kappa_{p} + 2\kappa_{m}} \right]$$
(3.2)

Eq. (3.2) shows that the migration velocity of DNA molecules in the globule state depends upon the difference of shape, volume, and polarizability among individual DNA molecules. The migration velocity of DNA molecules is in proportion to the square of the radius of DNA. So, it is possible to separate DNA molecules by length and polarizability by dielectrophoretic force. It is shown that the transportation of DNA with the state of compacted globule is profitable in the future practical application for the separation of giant DNAs such as human gene. In future we consider the Micro DNA Flow System and our new methodology utilizing conformational transition of higher order strucure of DNA as one of the biomanipulation and automation systems for DNA sequencing is a promising technique.

3.7 Summary

This chapter presented a new methodology on noncontact transportation of DNA molecules by dielectrophoretic force utilizing the conformational transition in the higher order structure of DNA for transportation. Experimental demonstration of DNA transportation in the globule state using dielectrophoretic force and direct observation of the DNA molecule in a nonuniform electric field were carried out with fluorescence microscopy. We show that transportation of DNA with the state of compacted globule is profitable in the future practical application for the separation of giant DNAs such as human gene.



Figure 3.8: Compacted globule DNA molecule suspending in a medium (at E=0)



Figure 3.9: Transportation trajectory that the DNA molecule migrates straight between the electrodes near X1 point (as shown in Fig. 3.7)



Figure 3.10: Transportation trajectory that the DNA molecule turns to the left near X2 point (as shown in Fig. 3.7)


Figure 3.11: Transportation trajectory that the DNA molecule turns to the right near X3 point (as shown in Fig. 3.7)



Figure 3.12: Transportation trajectory that the DNA molecule near X4 point (as shown in Fig. 3.7) migrates to the left toward the Ch2



Figure 3.13: Transportation trajectory that the DNA molecule near X4 point (as shown in Fig. 3.7) migrates to the right toward the Ch3



Figure 3.14: Transportation trajectory that the DNA molecule migrates in a zigzag through X4 point (as shown in Fig. 3.7) by switching the electric field (X: switching point)



Figure 3.15: Calculation of transportation trajectory of DNA molecule around the micropacket $(\delta 1, \delta 2) = (120^\circ, 30^\circ)$ from X1 point to X4 point



Figure 3.16: Calculation of transportation trajectory that the DNA molecule near X4 point migrates to the left toward the Ch2 in the micropacket $(\delta 1, \delta 2) = (120^{\circ}, 30^{\circ})$



Figure 3.17: Calculation of transportation trajectory that the DNA molecule near X4 point migrates to the right toward the Ch3 in the micropacket $(\delta 1, \delta 2) = (120^{\circ}, 30^{\circ})$

Chapter 4

Screening of a Single Escherichia coli in Microchannel by Electric Field and Laser tweezers

4.1 Introduction

In this chapter, we present a new high throughput screening method in microchannel. We have proposed the selective transportation system of DNA molecules for the purpose of transporting and separating a single DNA molecule by use of the dielectrophoretic force in Micro DNA Flow System [127, 128].

In this study, we propose a microsystem combined with optical trapping and dielectrophoretic force and carried out a preliminary experiment of manipulation of a single Escherichia coli. So, the technique, which isolates a single Escherichia coli speedingly and reliably, is needed. The methodology, that an arbitrary biological particles in the aggregation state can be manipulated, is needed because there are a number of Escherichia colis in an aqueous solution.

4.2 High Throughput Screening of Escherichia coli

4.2.1 Operation Improvement by Dielectrophoretic Force and Laser tweezers

Generally, many different kinds of forces are acting on the cell in solution such as the gravity, buoyancy, resistance force from viscosity of the fluid, Brownian motion, interactive forces in the micro/nano world such as Van der Waals force, electrostatic force depending on the surface charges. Moreover, if there is a flow, fluid force will disturb its motion. To control the biological object, we need to control the external force. To apply external force acting on the biological object, the following methods can be used.

(1) Electric Field

Coulomb force (D.C. field)

Dielectrophoretic force (A.C. field) [14, 37, 111, 112]

- (2) Ultrasonic Field [39,40]
- (3) Laser Beam [46, 47, 77, 78]
- (4) Fluid Flow of Solution

(2) and (3) require position information of the object, and the sensing system is needed for automation. However, when we consider precise and noninvasive handling of micro objects, (3) is suitable tool. (4) will disturb the environment with the fluid flow. So, we can simplify the system by using the electric field and laser tweezers. In case of using coulomb force, convection caused by electroosmosis, polarization of the microelectrode, and electrolysis will occur easily. So, we propose to use dielectrophoretic force to control the biological object.

4.2.2 Dielectrophoretic dispersion of Escherichia coli

In a high frequency field, the Escherichia coli is drawn in the direction of the high field gradient because the dielectrophoretic force operates in the direction of $qradE^2$. The force acting on the Escherichia coli in a nonuniform field is shown in Fig.4.1. Fig.4.2 shows the schematic diagram of the observation sample in the experiment. Fig.4.3 shows the schematic diagram of the electrode design. The force F_{die} and F_d , as shown in Fig.4.3, represent the dielectrophoretic force and the viscous force, when the high frequency electric field is applied between the electrode 3 and the electrode 4. In generating a high frequency AC field, the field around the Escherichia coli becomes a nonuniform field. Due to this electric field, the positive and negative charges in the Escherichia coli generate toward the electric field by its polarization. The polarizability of the Escherichia coli depends not only upon the dielectric characteristic, but also upon its conductivity. Some cells and Escherichia colis have conductivity in an aqueous solution. So, their conductivity as well as the dielectric characteristic is very important. As a result, the Escherichia colis migrate toward the normal to the line of electric line and aggregate in the state of the pearl chain formation by dielectrophoretic force. Once the target

Escherichia coli is trapped at the focal point of the laser beam, we can easily realize noncontact dispersion of unneeded Escherichia colis automatically by controlling the magnitude of the electric field.



Figure 4.1: Electrical dispersion of Escherichia coli by dielectrophoretic force

4.2.3 Optical Trapping of Escherichia coli by Laser Tweezers

Many biologists are eager to manipulate and separate a single biological particle for their experiment. With the conventional method, it takes so many times to reculture on plates. This method has such strong probability for pure culture of targeted microbes, such as a Escherichia coli. It will be such a promising method of pure culture of microbe for the biologists to discover a new microbe. It is possible to manipulate microbes, such as a Escherichia



Figure 4.2: Schematic diagram of the observation sample

coli and a yeast cell, with the Laser manipulator we used. Once a target Escherichia coli is trapped at the focal point of the laser beam, we can easily realize noncontact manipulation automatically.

4.3 Experiments

The cultured Escherichia colis diluted with a distilled water were centrifuged. We showed the transportation and dispersion of the Escherichia colis and the observation of the migration of the Escherichia colis when the high frequency electric field switches on and off. We carried out experiments as follows: the applied frequency is constant 1MHz and the applied voltage is constant 10 100Vp-p. By using the electrode samples, as shown in Fig.4.3, we observed the motion of the Escherichia coli by switching the electric field. We designed the electrode (as shown in Fig.4.3) for the purpose of transporting



Figure 4.3: Schematic diagram of the electrode design

the Escherichia coli. The thickness of the electrode is $40\mu m$. The minimum electrode gap is $50\mu m$.

The experimental apparatus is schematically shown in Fig.4.4. This laser manipulator consists of the inverted microscope and two high power laser diodes (wave length = 690nm). It is possible to manipulate two micro objects simultaneously. With one of the laser diodes, the focal point in a microscopic view can be moved arbitrarily with two degrees of freedom by controlling two galvano mirrors in the optical system. The trapped objects follows the focal point. The other laser diode is fixed, so the trapped objects cannot be moved without moving the XY stage. The resolution of the XY stage, which is controlled by micro stepping motor, is $1\mu m$. The observation of Escherichia coli has been carried out with a microscope, as shown in Fig.4.4, equipped with a 100 x oil-immersed objective lens and recorded on videotapes with the sampling rate of 30 frames/s through CCD camera. The applied voltage was controlled by an arbitrary waveform generator.



Figure 4.4: Schematic diagram of experimental apparatus

4.4 **Results and Discussion**

In our previous experiments [8-9], we observed the dynamics of DNA molecules between microelectrodes in an AC field. We showed the possibility of noncontact transportation of a DNA molecule by using dielectrophoretic force. We have confirmed the dependencies of the velocity of DNA molecule on the difference in the angle between electrodes, which is one of parameters of the electrode design. As the angle of the electrodes, where the electrode gap is the same size, becomes larger, the velocity of migration of a DNA molecule is faster. Because the angle of the electrodes is bigger, the difference in the field gradient is bigger.

From these results, we determined the observation sample design and carried out the experiments by using the observation sample chamber, as shown in Fig.4.2.

It is necessary to disperse unnecessary Escherichia colis toward each channel, as shown in Fig.4.2, in order to manipulate single Escherichia coli by laser tweezers at the center of the observation sample chamber.

Fig.4.5 Fig.4.7 show the motion of the Escherichia colis while the high frequency field switches on and off near the packet. They show that by controlling an electric field between electrodes, it is possible that the Escherichia colis are easily trapped by laser tweezers and transported and dispersed by high frequency electric field through the packet. Fig.4.5 illustrates the microscopic image of the handling of Escherichia coli in Microchannel by optical trapping while the high frequency field was off near the packet. Fig.4.6 and Fig.4.7 show the sequence images while the high frequency field switches on, and the pearl chain formation of Escherichia coli when the high frequency electric field generated between the microelectrodes. In each cases, Escherichia colis don't migrate when the electric field is off.

Fig.4.8 and Fig.4.9 show the motion of the Escherichia colis while the high frequency field switches on and off immediately near the packet. These sequence images depicted that the response of the movement of the Escherichia

73

colis to the high frequency field is very fast.

Fig.4.10 and Fig.4.11 show the motion of the Escherichia colis while the high frequency field switches on and off between the parallel electrodes. Between the parallel electrodes, the electric field is an uniform field. These images illustrates that the Escherichia colis don't migrate through the microelectrodes and make the pearl chain formation and keep the aggregation state between the parallel electrodes.

Fig.4.12 shows the pearl chain formation of the Escherichia colis near the entrance of channel 2 while the high frequency field switches on. They show that by controlling an electric field between electrodes, it is possible that Escherichia colis in aggregation state are easily separated through the packet.

4.5 Summary

In this chapter, we proposed a novel methodology on high throughput screening and micro surgery. We have developed a prototype of Microchannel system for high throughput screening of Escherichia coli. Experimental demonstration of Escherichia coli transportation using dielectrophoretic force and direct observation of the Escherichia coli in a non-uniform electric field were carried out with Laser manipulator system. We show that transportation and separation of Escherichia coli by dielectrophoretic force and optical trapping is profitable in the future practical application for the high throughput screening of microorganisms.



Figure 4.5: Microscope image of Escherichia coli trapped by Laser tweezers in an experiment, while the high frequency field switches off (E=0)



Figure 4.6: Sequence picture 1 that the Escherichia coli in the state of pearl chain formation between the electrode2 and the electrode3 migrate toward the Ch2, while the high frequency field switches on



Figure 4.7: Sequence picture 2 that the Escherichia coli in the state of pearl chain formation between the electrode2 and the electrode3 migrating toward the Ch2, while the high frequency field switches on



Figure 4.8: Microscope image of Escherichia coli trapped by Laser tweezers in an experiment, while the high frequency field switches off (E=0)



Figure 4.9: Transportation of the Escherichia coli in the state of pearl chain formation between the electrode2 and the electrode3 migrating toward the Ch2



Figure 4.10: Aggregation state of the Escherichia coli while the high frequency field switches off between the parallel electrodes



Figure 4.11: Aggregation state of the Escherichia coli in the state of pearl chain formation, while the high frequency field switches on between the parallel electrodes



Figure 4.12: Aggregation state of the Escherichia colis in the state of pearl chain formation near the entrance of channel 2 while the high frequency field switches on

Chapter 5

Bio-Micromanipulation System for High Throughput Screening and Micro Surgery in Microchannel

5.1 High Throughput Screening System of Microbes

Recently, there has been great interest in the high throughput screening and micro surgery of microorganisms for the finding of the novel microbes and the development of drug discovery process. Biotechnology has progressed in genetic engineering, life science, medicine and agriculture. For remarkable advancement of bioscience, it is important to develop the new technology, which manipulate and separate microorganisms with high speed and high purity and noninvasive method. In these research fields, operation of the micro/nano manipulation of cells and DNA, mass production, sequential processing, and high speed and high precision processing are required. However, the working efficiency and operability is dependent on the skills of the human operator. For the breakthrough in these fields, integration of the distributed research fields and system technologies is important. At this moment, micro electro mechanical systems technology and micro robotics has potentiality for the breakthrough in the bioengineering. For these multidisciplinary research areas, there are big challenges to improve the conventional method and develop the new system integration . We have considered the basic strategies to improve the operability of the bio micro manipulation. We should integrate the robotics and bioengineering. We call this research field as the bio-robotics [8].

Basic technologies of the bio-robotics are manipulation technology, micro system technology, visualization technology, human interface technology to improve operability, automation technology. Especially, micro/nano manipulation of a DNA molecule, animal or plant cell, and embryo and high throughput screening of microbes is important. However, research works is not enough at present.

5.2 Important Technologies for Bio Micro Manipulation

In this section, at first, we classify basic technologies required for the bio micro manipulation and we apply our methodology and strategies to high throughput screening of Escherichia coli in microchannel as follows:

- i) Separation, transportation, and position / orientation control technology
 Contact, non contact, High speed, high precision
- ii) MEMS technology

Fabrication, microchannel, micro valve, micro physics, micro sensor

84

iii) Measurement / control technology

Sensor, Image processing, teleoperation

iv) Visualization technology

Static, dynamic, 2D, 3D, fluorescence imaging

As an example of the contact tasks, nuclear transplantation or embryo culture requires those technologies. Most of the bio micro manipulation tasks are performed in an aqueous solution. Thus, compared with the general manipulation tasks in the air [6,7], we have to consider not only the gravitational force but also the buoyancy, fluid force from flow, Brownian motion, interactive forces in the micro/nano world such as Van der Waals force, electrostatic force depending on the surface charges. Moreover, the size of each object to manipulate is in micron order, it is quite difficult to recognize its configuration and to manipulate it freely. Automation of manipulation is quite difficult, so the skills of the human operator are required. From these point of view, we summarize the basic strategies to improve the working efficiency and the operability of the bio micro manipulation for high throughput screening and micro surgery of microbes as follows:

i) Utilization of optical trapping by the laser tweezers [46, 47, 77, 78]

At present, polylysine or micro pipette is used for fixation. Fixation by the polylysine is easy, but this method is not reversible. Automation of the micro pipette operation is difficult. So, we need to develop a new method in a different way. Once the biological object is trapped

85

at the focal point of the laser beam, we can easily realize noncontact manipulation automatically. In this chapter, we show the system flow diagram utilization of optical trapping by the laser tweezers.

 ii) Utilization of dielectrophoretic force generated between the microelectrodes [14, 37, 111, 112]

To improve working efficiency and operability in solution and to realize high speed separation, the technique for sequential processing is needed. Once the object is trapped at the focal point of the laser beam, we can easily realize noncontact dispersion of unneeded microbes automatically by controlling the magnitude of the electric field.

iii) Realization of autonomous function

Micro manipulation tasks are versatile. So, it is difficult to realize a full automation system. We should start from classifying the basic operation for high throughput screening. Most of the contact manipulation, such as nuclear transplantation, is performed by the human operator by hand with the microscope. We cant expect quick response nor precise and repetitive operation by hand. So, we should realize shearing of the autonomous function and manual operation.

5.3 New methodology of selective transportation system of Escherichia coli

We have developed a prototype of Microchannel system (Fig.5.1) for high throughput screening of Escherichia coli. Experimental demonstration is shown.



Figure 5.1: Micro Glass Chip for High Throughput Screening of Microbes

Fig.5.2 shows the sequence flow chart and the system flow diagram for our proposed high throughput screening of Escherichia coli. At first, in order to specify a target Escherichia coli, we propose the detection of the target Escherichia coli by image recognition. The characteristics of each Escherichia coli is dependent on the fluorescent color, the shape, the mobility, and the migration ability etc. Using the fluorescence microscopy, it is possible to detect the fluorescence image of the Escherichia coli. The sequence flow chart we propose is as follows:



Figure 5.2: System flow diagram for high throughput screening of Escherichia coli

- i) Image recognition of a target Escherichia coli by Image Analyzer
- ii) Setting the coordinates which the laser beam radiates by controlling of XY Stage or galvano mirror
- iii) Laser beam ON
- iv) Trapping of the target Escherichia coli by laser tweezers
- v) Dispersion of unneeded Escherichia colis in an aqueous solution by dielectrophoretic force while controlling of the magnitude of the electric field

- vi) Recovery of single Escherichia coli
- vii) Finally, a single Escherichia coli is separated.



Figure 5.3: Schematic diagram of Fluorescence Microscopy with Laser Manipulator

In this chapter, we propose new methodology of selective transportation system of Escherichia coli by dielectrophoretic force and optical radiation pressure and carried out a preliminary experiment. We demonstrated the experiments through the sequence flow chart (2) to (5), as shown in Fig.5.2.

5.4 Application of Liposome to Micro Encapsulation Delivery of Escherichia coli

5.4.1 Giant Liposome as Micro Capsule

Liposomes are the simplest artificial structures, resembling natural membranedelimited domains [85, 88, 89]. Fig.5.4 shows the schematic diagram of liposome. When Escherichia coli and DNA molecule are entrapped in liposomes [90,91], the resulting supramolecular structures exhibit very interesting properties from both a fundamental and a practical point of view. Fig.5.5 shows the concept of trapping single globular DNA molecule by Laser tweezers and its application to micro encapsulation by liposome. DNA molecule-loaded liposomes can be regarded as simplified models of biomembranes and they can be used as highly specific microreactors for exploiting the genetic information in a variety of applications. This compartmentation contributes significantly to the entrapment of microbes and single globular DNA molecule and to the optimization of the cell behaviour.

5.4.2 Optical Trapping of Liposome by Laser tweezers

Neutral phospholipids were obtained from Sigma, and inclueded soybean phosphatidylcholine (soybean PC) and dioleoylphosphatidyl-choline (DOPC). These lipids were dissolved in pure methanol or in a 1 : 1 (v : v) chloroformmethanol mixture. Next, $0.1\mu mol$ of dry lipid film was prepared by evaporation of the organic solvent in a $10 - mm\phi$ test tube under a nitrogen stream, and then stored for at least 2 h under aspiration. The resulting lipid film



Figure 5.4: Schematic diagram of Liposome

was swollen with $100\mu ml$ of TBE buffer (90mM boric acid, 2mM EDTA, pH adjusted to 8.0 by HCl) or aqueous solution containing $MgCl_2$, where the final concentration of phospholipid was 1mM. Fig.5.6 shows microscopic image of liposome in Microchannel trapped by Laser tweezers. We can control the conformation of liposome and apply it to the micro encapsulation delivery of microbe.

5.5 Summary

In experiments, we show that transportation and separation of Escherichia coli and Liposomes with electric field and optical radiation pressure. Isolation of Escherichia coli by fluorescence microscopy with Laser Manipulator was successful. Using this system, it is profitable in the future practical application for the high throughput screening of microbes and drug delivery and drug discovery.



Figure 5.5: Concept of trapping Single globular DNA by Laser tweezers and its Application to Micro encapsulation by Liposome



Figure 5.6: Microscopic Image of Liposome trapped by Laser tweezers

Chapter 6 Conclusion

6.1 Summary

In this chapter, I summarize the results obtained in the previous chapters systematically.

Chapter 2 described experiments involving noncontact transportation of DNA molecules by using dielectrophoretic force. Compared with the method employing electrophoresis, electroosmosis and chemical reaction on electrodes doesn't occur by use of an AC field. In a Micro DNA Flow System it is also easier to extend the flow channel. We designed a simple electrode, carried out the experiments and measured the velocity of migration of the DNA molecule. We observed the dynamics of DNA molecules between electrodes in an AC field. We showed the possibility of noncontact transportation of a DNA molecule by using dielectrophoretic force.

In Chapter 3, we demonstrated the transportation of DNA molecule in the globule state by dielectrophoretic force. A DNA molecule is polyelectrolyte. However, the compactization of a single DNA molecule is considered to induce charge neutralization, so it is not effective to use the electrophoresis that is one of the methods of separation of charged particles. Dielectrophoresis is suitable for transportation and separation of dielectric particles. In the experiments, we observed the transportation of the DNA molecule in the globule state stably. We show that transportation of DNA with the state of compacted globule is profitable in the future practical application for the separation of giant DNAs such as human gene.

Chapter 4 proposed a novel methodology on high throughput screening of Escherichia coli. Experimental demonstration of Escherichia coli transportation using dielectrophoretic force and direct observation of the Escherichia coli in a non-uniform electric field were carried out with Laser manipulator system. By controlling an electric field between electrodes, it is possible that Escherichia colis in aggregation state are easily separated through the packet. We showed that transportation and separation of Escherichia coli by dielectrophoretic force and optical trapping is profitable in the future practical application for the high throughput screening of microorganisms.

Chapter 5 proposed a novel methodology and system on high throughput screening of microbes. We have developed a prototype of Microchannel system for high throughput screening of Escherichia coli. We discussed research issues in the bio micro manipulation and presented a new direction in this field. We consider the basic strategies to improve the working efficiency and the operability of the bio micro manipulation. In experiments, we show that transportation and separation of Escherichia coli with electric field and optical radiation pressure, and application of liposome to micro encapsulation delivery

95

is profitable in the future practical application for the high throughput screening of microbes. We discuss the application of liposome to micro encapsulation delivery of biomaterials.

6.2 Future work

For these multidisciplinary research areas of biotechnology, there are big challenges to improve the conventional method and develop the new system integration for Bio-Micromanipulation. We have to consider the basic strategies to improve the operability of the bio micro manipulation. We should integrate the robotics and bioengineering.

For example, in the case of DNA sequencing, determining the human genome sequence and finding the genes is really just a first step. After sequencing, we still need to determine what proteins the genes produce, and what those proteins do in the cell. We still need to know the structure and function of the protein produced by the gene, and how that protein interacts in the environment of the cell. Bioinformatics is a necessary field in these studies. When we design the system for a various kinds of research in bioscience field, we have to continue to consider system integration and new methodology for the progress of lifescience and bioscience.
References

- T. Fukuda, K. Tanie, T. Mitsuoka: A Study on Control of a Micromanipulator (1st Report, The basic feature of Micro Gripper and one method of Bilateral control), Proc. Micro Robots and Teleoperators Workshop, Hyannis, MA, Nov. 1987.
- [2] T. Fukuda, O. Hasegawa: Creature Recognition and Identification by Image Processing Based on Expert System, IEEE Int'l Conf. on SMC., 837-842, 1989.
- [3] T. Fukuda, T. Tanaka: Micro Electro Static Actuator with Three Degrees of Freedom, Proc. IEEE International Workshop on MICRO ELECTRO MECHANICAL SYSTEMS, 153-158, February 1990.
- [4] T. Fukuda, M. Fujiyoshi, F. Arai, H. Matsuura: Control of Manipulator (4st Report, Design and Experimental Result of Micromanipulator with 6 D. O. F), Trans. of JSME, Vol. 56, No. 532, 3287-3293, 1990 (in Japanese).
- [5] T. Fukuda, M. Fujiyoshi, F. Arai, H. Matsuura: Control of Manipulator (7st Report, Design and Experimental Result of Micromanipulator with Electrostatic Actuators), Trans. of JSME, Vol. 57, No. 536, 1296-1301, 1991 (in Japanese).
- [6] F. Arai, D. Andou, T. Fukuda, Y. Nonoda, T. Ota: Micro Manipulation Based on Micro Physics -Strategy Based on Attractive Force Reduction and Stress Measurement-, Proc. Int. Conf. on Intelligent Robotics and Systems (IROS), Vol. 2, 236-241, 1995.
- [7] F. Arai, D. Andou, Y. Nonoda, T. Fukuda, H. Iwata, K. Itoigawa: Micro Endeffector with Micro Pyramids and Integrated Piezoresistive Force Sensor, Proc. Int. Conf. on Intelligent Robotics and Systems(IROS), Vol. 2, 842-849, 1996.
- [8] F. Arai, K. Morishima, T. Kasugai, T. Fukuda: Bio-Micro-Manipulation (New Direction for Operation Improvement), Proc. Int. Conf. on Intelligent Robotics and Systems(IROS), Vol.3, 1300-1305, 1997-9.
- [9] J. W. Jorgenson, K. D. Lukacs: Free-zone electrophoresis in glass capillaries, Clinical Chemistry, Vol. 27, No. 9, 1551-1553, 1981

- [10] R. Hagedorn, G. Fuhr: Calculation of rotation of biological objects in the electric rotating field, Stud. biophys., Vol. 102, No. 3, 229-238, 1984.
- [11] G. Fuhr, R. Hagedorn, H. Göring: Separation of different cell types by rotating electricfields, Plant Cell Physiol., Vol. 26, No. 8, 1527-1531, 1985.
- [12] G. Fuhr, P. I. Kuzmin: Behaviour of cells in rotating electric fields with account to surface charges and cell structure, Biophys. J, 50, 789-795, 1986.
- [13] G. Fuhr, P. Rösch, T. Müller, V. Dressier, H. Göring: Dielectric spectroscopy of chloroplasts isolated from higher plants -characterization of the double -membrane system, Plant Cell Physiol., Vol. 31, No. 7, 975-985, 1990.
- [14] G. Fuhr, R. Hagedorn, T. Müller, B. Wanger, W. Benecke: Linear motion of dielectric particles and living cells in microfabricated structures induced by travelling electric fields, Proceeding of IEEE Micro Electro Mechanical Systems, 259-264, 1991.
- [15] R. Hagendorn, G. Fuhr, T. Müller, J. Gisma: Travering-wave dielectrophoresis(TWD)of microparticles, Electrophoresis, Vol. 13, 49-54, 1992.
- [16] G. Fuhr, W. M. Arnold, R. Hagendorn, T. Müller, W. Benecke, B. Wanger, U. Zimmermann: Levitation, holding and rotation of cells within traps made by high-frequency fields, Biochim. Biophys. Acta, Vol. 1108, 215-223, 1992.
- [17] G. Fuhr, R. Hagendorn, T. Müller, W. Benecke, B. Wanger: Microfabricated electrohydrodynamic(EHD)pumps for liquids of higher conductivity, J. Microelectromech. Systems, Vol. 1, No. 3, 141-146, 1992.
- [18] Th. Schnelle, R. Hagendorn, G. Fuhr, St. Fledler, T. Müller: Threedimensional electric field traps for manipulation of cells-calculation and experimental venification, Biochim. Biophys. Acta, Vol. 1157, 127-140, 1993.
- [19] G. Fuhr, T. Müller, Th. Schnelle, R. Hagedorn, A. Voigt, S. Fiedler: Radio-Frequency Microtools for Particle and Live Cell Manipulation, Naturwissenschaften, Vol. 81, 528-535,1994.
- [20] G. Fuhr, St. Fledler, T. Müller, Th. Schnelle, H. Glasser, T. Lisec, B. Wanger: Particle micromanipulator consisting of two orthogonal channeles with traveling-wave electrode structures, Sensors and Actuators A41-42, 230-239, 1994.
- [21] G. Fuhr, H. Glasser, T. Müller, Th. Schnelle: Cell manipulation and cultivation under a.c. electric field influence in highly conductive culture media, Biochim. Biophys. Acta, Vol. 1201, 353-360, 1994.

- [22] G. Fuhr, G. S. Shirley: Cell handling and caracterization using micron and submicron electrode arrays :state of the art and perspectives of semiconductor microtools, J. Michromech. Microeng., Vol. 5, 77-85, 1995.
- [23] G. Fuhr, Th. Schnelle, T. Müller, Th. Schnelle, H. Glasser, Th. Lisec, B. Wanger: Positioning and manipulation of cells and microparticles using miniaturized electric field traps and travelling waves, Sensors and Materials, Vol. 7, No. 2, 131-146, 1995.
- [24] G. Fuhr, Th. Schnelle, S. G. Shiriey, R. Hagendorn: Dielectrophoretic field-cages-a new technique for cell, virus and macromolecule handling, Cellular Engineering, 1, 47-57, 1995.
- [25] G. Fuhr, A. Voigt, T. Müller, B. Wanger, K. Reimer, T. Lisec: Electric field-mediated inhibition of cell and microparticle adhesion: a new way to create bio-repellent surface, Sensors and Actuators B26-27, 468-470, 1995.
- [26] T. Müller, A. Gerardio, Th. Scnelle, G. S. Shirley, F. Bordoni, G. De-Gasperis, R. Leoni, G. Fuhr: Trapping of micron and submicron particles by high frequency fields and hydrodynamic forces, J. of Physics., Vol. 29, 340-349, 1996.
- [27] M. Ridely: Evolution, Blackwell Scientific Publications, 23-26, 1993.
- [28] From Genes to Genome Biology, Reserch News, SCIENCE, Vol. 272, 21 June, 1996
- [29] K. Yoshikawa, Y. Matsuzawa, K. Minagawa, M. Doi, M. Matsumoto: Opposite Effect between Intercalator and Minor Groove Binding Drug on the Higher Order Structure of DNA as Is Visualized by Fluorescence Microscopy, Biochem. Biophys. Res. Commun., Vol. 188, 1274-1279, 1992.
- [30] M. Matsumoto, T. Sakaguchi, M. Kimura, M. Doi, K. Minagawa, Y. Matsuzawa, K. Yoshikawa: Direct Observation of Brownian Motion of Macromolecules by Fluorescence Microscopy, J. Polym. Sci., PartB: Polym. Phys., Vol. 30, 779-783, 1992.
- [31] H. Oana, Y. Masubuchi, M. Matsumoto, M. Doi, K. Minagawa, Y. Matsuzawa, K. Yoshikawa: Periodic Motion of Large DNA Molecules during Steady Field Gel, Macromolecules, Vol. 27, 6061-6067, 1994.
- [32] S. M. Mel'nikov, V. G. Sergeyev, K. Yoshikawa: Discrete Coil-Globule Transition of Large DNA Induced by Cationic Surfactant, J. Am. Chem. Soc., Vol. 117, No. 9, 2401-2408, 1995.
- [33] H. Oana, K. Yoshikawa: Dynamics of the Higher Order Structure of Giant DNA-Single Molecular Observation by Fluorescene Microscopy, Protein, Nucleic Acid and Enzyme Vol. 40, No. 10, 1558-1563, 1995, in Japanese.

- [34] V. V. Vasilevskaya, A. R. Khokhlov, Y. Matsuzawa, K. Yoshikawa: Collapse of Single DNA in poly (ethylene glycol) solutions, J. Chem. Phys., Vol. 102, No. 16, 6595-6602, 1995.
- [35] M. Ueda, K. Yoshikawa: Phase Transition and Phase Segregation in a Single Double-Stranded DNA Molecule, Physical Review Letters, Vol. 77, No. 10, 2133-2136, 1996.
- [36] Y. Matsuzawa, K. Yoshikawa: Conformational Transition of Higher-Order Structure in Single DNA Molecule, Kobunshi Ronbunshu, Vol. 52, No. 12, 753-764, 1995, in Japanese.
- [37] H. A. Pohl: Dielectrophoreshis, Cambridge University Press, Cambridge, 1-579, 1978.
- [38] T.L.Mahaworasilpa, H. G. L. Coster, E. P. George: Forces on biological cells due to applied alternating(AC) electric fields, Biochimica Biophysica Acta, 1193, 118-126, 1994.
- [39] H. Mitome: Control of Force in a Micro Region Using Acoustic Radiation Pressure, 4th International Symposium on Micro Machine and Human Science, 68-74, 1993.
- [40] E. Benes: Trapping of Suspended Biological Particles by use of Ultrasonic Resonance Fields, Ultrasonics International 93 Abstract, 39, 1993.
- [41] X-B Wang: Selective Dielectrophoretic Confinement of Bioparticles in Potential Energy Wells, J. Phys. D:Appl. Phys. 26, 1278-1285, 1993.
- [42] K. Morishima, T. Fukuda, F. Arai, H. Ishihara, H. Matsuura, K. Yoshikawa: Noncontact Transportation of DNA Molecule by DielectrophoreticForce, Proc. of IEEE 6th International Symposium on Micro Machine and Human Science, 145-152, 1995.
- [43] K. Morishima, T. Fukuda, F. Arai, H. Matsuura, K. Yoshikawa: Noncontact Transportation of DNA Molecule by Dielectrophoretic Force for Micro DNA Flow System, Proceedings 1996 IEEE International Conference on Robotics and Automation, Minneapolis, 2214-2219, 1996.
- [44] A. Mizuno, M. Nishioka, Lusian-Doru. Dascalescu: IEEE Transactions on Industry Applications, Vol. 31, No. 3, 464, 1995.
- [45] A. Mizuno, M. Nishioka, Lucian-Doru. Dascalescu: Handling of a Single DNA Molecule Using Electric Field and Laser Beam, IEEE Transaction on Industry Applications, Vol. 31, No. 6, 1452,1995.
- [46] A. Mizuno, M. Nishioka, R. Ishii, S. Katsura: Opto-Electrostatic Micromanipulation of Single Cell and DNA Molecule, Proc. of IEEE 6th International Symposium on Micro Machine and Human Science, 153-159, 1995.

- [47] A. Mizuno, M. Nishioka, R. Ishii, S. Katsura: Opto-Electrostatic Micromanipulation of Single Cell and DNA Molecule, Proc. of IEEE 6th International Symposium on Micro Machine and Human Science, 153-159, 1995.
- [48] S. C. Jacobson, R. Hergenröder, L. B. Koutny, J. M. Ramsey: High-Speed Separation on a Microchip, Anal. Chem., Vol. 66, 1114-1118, 1994.
- [49] S. C. Jacobson, A. W. Moore, J. M. Ramsey: Fused Quartz Substrates for Microchip Electrophoresis, Anal. Chem., Vol. 67, 2059-2063, 1995.
- [50] A. W. Moore, Jr., S. C. Jacobson, J. M. Ramsey: Microchip Separations of Neutral Species via Micellar Electrokinetic Capillary Chromatography, Anal. Chem., Vol. 67, 4184-4189, 1995.
- [51] S. C. Jacobson, J. M. Ramsey: Integrated Microdevice for DNA Restriction Fragment Analysis, Anal. Chem., Vol. 68, 720-723, 1996.
- [52] S. C. Jacobson, J. M. Ramsey: Electrokinetic Focusing in Microfabricated Channel Structures, Anal. Chem., Vol. 69, 3212-3217, 1997.
- [53] D. J. Harrison, A. Manz, Z. Fan, H. Lüdi, H. M. Widmer: Capillary Electrophoresis and Sample Injection Systems Integrated on a Planar Glass Chip, Anal. Chem., Vol. 64, 1926-1932, 1992.
- [54] D. J. Harrison, K. Fluri, K. Seiler, Z. Fan, C. S. Effenhauser, A. Manz: Micromachining a Miniaturized Capillary Electrophoresis-Based Chemical Analysis System on a Chip, SCIENCE, Vol. 261, 13 August, 1993.
- [55] Z. H. Fan, D. J. Harrison: Micromachinig of Capillary Electrophoresis Injectors and Separators on Glass Chips and Evaluation of Flow at Capillary Intersections, Anal. Chem., Vol. 66, 177-184, 1994.
- [56] K. Fluri, G. Fitzpatrick, N. Chiem, D. J. Harrison: Integrated Capillary Electrophoresis Devices with an Efficient Postcolumn Reactor in Planar Quartz and Glass Chips, Anal. Chem., Vol. 68, 4285-4290, 1996.
- [57] K. Seiler, D. J. Harrison, A. Manz: Planar Glass Chips for Capillary Electrophoresis: Repetitive Sample Injection, Quantation, and Separation Efficiency, Anal. Chem., Vol. 65, 1481-1488, 1993.
- [58] C. S. Effenhauser, A. Manz, H. M. Widmer: Glass Chips for High-Speed Capillary Electrophoresis Separations with Submicrometer Plate Heights, Anal. Chem., Vol. 65, 2637-2642, 1993.
- [59] C. S. Effenhauser, A. Paulus, A. Manz, H. M. Widmer: High-Speed Separation of Antisense Oligonucleotides on a Micromachined Capillary Electrophoresis Device, Anal. Chem. ,Vol. 66, 2949-2953, 1994.

- [60] The Incredible Shrinking Laboratory, Reserch News, SCIENCE, Vol. 268, 7 April, 1995.
- [61] A. T. Woolley, R. A. Mathies: Ultra-high-speed DNA fragment separations using microfabricated capillary array electrophresis chips, Proc. Natl. Acad. Sci. USA, Vol. 91,11348-11352, Nov, 1994.
- [62] A. T. Woolley, R. A. Mathies: Ultra-High-Speed DNA Sequencing Using Capillary Electrophoresis Chips, Anal. Chem. 67, 3676-3680, 1995.
- [63] A. T. Woolley, D. Hadley, P. Landre, A. J. deMello, R. A. Mathies, M. A. Northrup: Functional Integration of PCR Amplification and Capillary Electrophoresis in a Microfabricated DNA Analysis Device, Anal. Chem. , Vol. 68, 4081-4086, 1996.
- [64] C. S. Foong, K. L. Wood, I. Busch-Vischniac: Design Assessment of Micro-Electro-Mechanical Systems with Applications to a Microbiology Cell injector, Micromechanical Systems ASME, DSC-Vol.46, 49-63, 1993.
- [65] H. Swerdlow, B. J. Jones, C. T. Wittwer: Fully Automated DNA Reaction and Analysis in a Fluid Capillary Instrument, Anal. Chem., Vol. 69, 848-855, 1997.
- [66] D. R. Meldrum: A Biomechatronic Fluid Sample Handling System for DNA Processing, Proc. of Advanced Intelligent Mechatronics, 1997.
- [67] P. F. Gavin, A. G. Ewing: Continuous Separation with Microfabricated Electrophoresis-Electrochemical Array Detection, J. Am. Chem. Soc. ,118, 8932-8936,1996.
- [68] T. Anazawa, S. Takahashi, H. Kambara: A Capillary Array Gel Electrophoresis System Using Multiple Lazer Focusing for DNA Sequencing, Anal. Chem., Vol. 68, 2699-2704, 1996.
- [69] S. Mouradian, J. W. Skogen, F. D. Dorman, F. Zarrin: DNA Analysis Using an Electrospray Scanning Mobility Particle Sizer, Anal. Chem., Vol. 69, 919-925,1997.
- [70] R. M. McCormick, R. J. Nelson, M. G. Alonso-Amigo, D. J. Benvegnu, H. H. Hooper: Microchannel Electrophoretic Separations of DNA in Injection-Molded Plastic Substrates, Anal. Chem., Vol. 69, 2626-2630, 1997.
- [71] D. Figeys, Y. Ning, R. Aebersold: A Microfabricated Device for Rapid Protein Identification by Microelectrospray Ion Trap Mass Spectrometry, Anal. Chem., Vol. 69, 3153-3160, 1997.
- [72] Scot C. Kuo, Michael P. Sheetz: Force of Single Kinesin Molecules Measured with Optical Tweezers, SCIENCE Vol. 260, 232-234, 1993.

- [73] T. Imasaka, Y. Kawabata, T. Kaneta, Y. Ishidzu: Optical Chromatography, Anal. Chem., Vol. 67, 11,1 June,1995.
- [74] Steven Chu: Laser Manipulation of Atoms and Particles, SCIENCE, Vol. 253, 1991.
- [75] Steven Chu: Laser Trapping of Neutral Particles, SCIENTIFIC AMERI-CAN, Feb., 1992.
- [76] A. Ashkin: The Pressure of Laser Light, SCIENTIFIC AMERICAN, Vol. 226, No. 2, 1972.
- [77] A. Ashkin, J. M. Dziedzic: Optical trapping and manipulation of viruses and bacteria, Science, Vol. 235, 1517-1520, 1987.
- [78] A. Ashkin, J. M. Dziedzic, T. Yamane: Optical trapping and manipulation of single cells using infrared laser beams, Nature, Vol. 330, 769-771, 1987.
- [79] W. D. Volkmuth, R. H. Austin: DNA electrophoresis in microlithographic arrays, NATURE, Vol. 358, 13 August, 1992.
- [80] W. D. Volkmuth, T. Duke, M. C. Wu, R. H. Austin: DNA Electrodiffusion in a 2D Array of Posts, The American Physical Society, Vol. 72, No. 13, 28 March, 1994.
- [81] L. Mitnik, C. Heller, J. Prost, J. L. Viovy: Segregation in DNA Solutions Induced by Electric Fields, SCIENCE, Vol. 267, 13 Jan., 1995.
- [82] T. A. J. Duke, R. H. Austin, E. C. Cox, S. S. Chan: Pulsed-field Electrophoresis in Microlithographic arrays, Electrophoresis, 17, 1075-1079, 1996.
- [83] Carlos Bustamante: Direct Observation and Manipulation of Single DNA Molecules Using Fluorescence Microscopy, Annu. Rev. Biophys. Chem. 20, 415-46, 1991.
- [84] D. C. Schwartz, X. Li, L. I. Hernandez, S. P. Ramnarain, E. J. Huff, Y-K. Wang: Ordered Restriction Maps of Sccharomyces cerevisiae Chromosomes Constructed by Optical Mapping, SCIENCE, Vol. 262, 110-114, 1 October, 1993.
- [85] C. Fred. Fox: The Structure of Cell Membranes, SCIENTIFICAMERI-CAN, Vol. 226, No. 2, 1972.
- [86] D. Leung, D. D. Lasic: Stability and Stabilization of Liposome, Handbook of Nonmedical Applications of Liposomes, From Design to Microreactors, Edited by Y. Barenholz and D. D. Lasic, Vol. 3, 31-41, CRC Press, 1995.

- [87] A. T. Florence, T. Yoshioka: Nonionic Surfactant Vesicle-in-Water-in-Oil(v/w/o)Systems, Handbook of Nonmedical Applications of Liposomes , From Design to Microreactors, Edited by Y. Barenholz and D. D. Lasic, Vol. 3, 199-208, CRC Press, 1995.
- [88] Danilo D. Lasic: Liposome in Gene Therapy, Handbook of Nonmedical Applications of Liposomes, From Gene Delivery and Diagnostics to Ecology, Edited by D. D. Lasic and Y. Barenholz, Vol.4, 1-5, CRC Press, 1995.
- [89] H. Farhood, L. Huang: Delivery of DNA, RNA, and Proteins by Cationic Liposome, Handbook of Nonmedical Applications of Liposomes, From Gene Delivery and Diagnostics to Ecology, Edited by D. D. Lasic and Y. Barenholz, Vol.4, 31-42, CRC Press, 1995.
- [90] G.Gregoriadis, I. Gursel, S. G. Antimisaris: Microbe Entrapment in Giant Lipospmes, Handbook of Nonmedical Applications of Liposomes, From Gene Delivery and Diagnostics to Ecology, Edited by D. D. Lasic and Y. Barenholz, Vol.4, 263-269, CRC Press, 1995.
- [91] H. Miyata, H. Hotani: Morphological changes in liposomes caused by polymerization of encapsulated actin and spontaneous formation of actin bundles, Proc. Natl. Acad. Sci. USA, Vol. 89, 11547-11551, December, 1992.
- [92] N. G. Stoicheva, S. W. Hui: Electrofusion of cell-size liposomes, Biochimica et Biophysica Acta, 1195, 31-38, 1994.
- [93] N. G. Stoicheva, S. W. Hui: Dielectrophoresis of cell-size liposomes, Biochimica et Biophysica Acta, 1195, 39-44, 1994.
- [94] S. W. Hui, N. Stoicheva, Y-L. Zhao: High-Efficiency Loading, Transfection, and Fusion of Cells by Electroporation in Two-Phase Polymer Systems, Biophysical Journal, Vol.71, 1123-1130, August, 1996.
- [95] Chester t. O'Konski: Electric Properties of Macromolecules. V. Theory of Ionic Polarization in Polyelectrolytes, 605-619, May, 1960.
- [96] Gerald S. Manning: Limiting Laws and Counterion Condensation in Polyelectrolyte Solutions-The Approach to the Limit and the Extraordinary Stability of the Charge Fraction-, Biophysical Chemistry, Vol. 7, 95-102, 1977.
- [97] Gerald S. Manning: A Field-Dissociation Relation for Polyelectrolymes with an Application to Field-Induced Conformational Changes of Polynucleotides, Biophysical Chemistry, Vol. 9, 189-192, 1977.
- [98] J. C. W. Shehpherd, G. Schwarz: Dielectric Studies on Self-Associating Nucleosides and Bases in Aqueous Solution, Biophysical Chemistry, Vol. 7, 193-204, 1977.

- [99] S. W. Lee, J. H. Choi, Y. K. Kim: Analysis and experimental investigation of dielectrophoretic force in a planar electrode structure, J. Micromech. Microeng., 6, 66-68, 1996.
- [100] R. Pethig, Y. Huang, X. B. Wang, J. P. H. Burt: Positive and negative dielectrophoretic collection of colloidal particles using interdigited castellated microelectrodes, J. Phys. D: Appl. Phys., 24, 881-888, 1992.
- Y. Huang, X. B. Wang, J. A. Tame, R. Pethig: Electrokinetic behaviour of colloidal particles in travelling electric fields: studies using yeast cells, J. Phys. D: Appl. Phys., 26, 1528-1535, 1993.
- [102] X. B. Wang, M. P. Hughes, Y. Huang, F. F. Becker, P. R. C. Gascoyne: Non-uniform spatial distributions of both the magnitude and phase of AC electric fields determine dielectrophoretic forces, Biochimica et Biophysica Acta, 1243, 185-194, 1995.
- [103] M. P. Hughes, R. Pethig, X. B. Wang: Dielectrophoretic forces on particles in travelling electric fields, J. Phys. D: Appl. Phys. , 29, 474-482, 1996.
- [104] X. Wang, X. B. Wang, F. F. Becker, P. R. C. Gascoyne: A theoretical method of electrical field analysis for dielectrophoretic electrode arrays using Green's theorem, J. Phys. D: Appl. Phys., 29, 1649-1660, 1996.
- [105] M. S. Talary, J. P. H. Burt, J. A. Tame, R. Pethig: Electromanipulation and separation of cells using traveling electric fields, J. Phys. D: Appl. Phys. ,29, 2198-2203, 1996.
- [106] G. H. Markx, R. Pethig, J. Rousselet: The dielectrophoretic levitation of latex beads, with reference to field-flow fractionation, J. Phys. D: Appl. Phys., 30, 2470-2477, 1997.
- [107] X. B. Wang, Y. Huang, X. Wang, F. F. Becker, P. R. C. Gascoyne: Dielectrophoretic Manipulation of Cells with Spiral Electrodes, Biophysical Journal, 72, 1887-1899, 1997.
- [108] P. R. C. Gascoyne, J. Noshari, F. F. Becker, R. Pethig: Use of Dielectrophoretic Collection Spectra for Characterizing Differences between Normal and Cancerous Cells, IEEE Tran. Industry Applicat., Vol. 30, No. 4, 829-834, 1994.
- [109] R. Hölzel: Electrorotation of Single Yeast Cells at Frequencies Between 100 Hz and 1.6 GHz, Biophysical Journal, 73, 1103-1109, 1997.
- [110] S. Masuda, M. Washizu, I. Kawabata: Movement of Blood Cells in Liquid by Nonuniform Travelling Field, IEEE Tran. Industry Applicat., Vol. 24, No. 2, 217-222, 1988.

- [111] S. Masuda, M. Washizu, T. Nanba: Novel Method of Cell Fusion in Field Constriction Area in Fluid Integrated Circuit, IEEE Tran. Industry Applicat., Vol. 25, No. 4, 732-737, 1989.
- [112] M. Washizu, T. Nanba, S. Masuda: Handling Biological Cells Using a Fluid Integrated Circuit, IEEE Tran. Industry Applicat., Vol. 26, No. 2, 352-358, 1990.
- [113] M. Washizu: Handling of Biological Molecules and Membranes in Microfabricated Structure, Automation in Biotechnology(The 4th TOYOTA Conference), Oct., 14-1-14-13, 1990.
- [114] M. Washizu, O. Kurosawa: Electrostatic Manipulation of DNA in Microfabricated Structures, IEEE Tran. Industry Applicat., Vol. 26, No. 6, 1165-1172, 1990.
- [115] M. Washizu: Manipulation of Biological Objects in Micromachined Structures, Proceeding of IEEE Micro Electro Mechanical Systems, 196-201, 1992.
- [116] M. Washizu, S. Suzuki, O. Kurosawa, T. Nishizaka, T. Shinohara: Molecular Dielectrophoresis of Biopolymers, IEEE Tran. Industry Applicat., Vol. 30, No. 4, 835-843, 1994.
- [117] N. G. Green, H. Morgan: Dielectrophoretic separation of nano-particles, J. Phys. D: Appl. Phys., 30, L41-L44, 1997.
- [118] B. Prüger, P. Eppmann, E. Donath, J. Gimsa: Measurement of Inherent Particle Properties by Dynamic Light Scattering: Introducing Electrorotational Light Scattering, Biophysical Journal, Vol. 72, 1414-1424, March, 1997.
- [119] Y. Huang, X-B. Wang, F. F. Becker, P. R. C. Gascoyne: Introducing Dielectrophoresis as a New Force Field for Field-Flow Fractionation, Biophysical Journal, Vol. 73, 1118-1129, August, 1997.
- [120] T. Fukuda, K. Morishima, F. Arai, H. Matsuura, K. Yoshikawa: Noncontact Manipulation of DNA Molecule (1st Report, Transportation of DNA Molecule by Dielectrophoretic Force), Trans. of JSME, Vol. 62, No. 599, 249-256, 1996-7 (in Japanese).
- [121] K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Transportation of DNA Molecule by Dielectrophoretic Force in Micro Flow Channel -Utilization of Conformational Transition in the Higher Order Structure of DNA-, Trans. IEE of Japan, Vol.116-E, No.9, 387- 394, Nov., 1996.
- [122] K. Morishima, F. Arai, T. Fukuda, H. Matsuura, K. Yoshikawa: Screening of Single Escherichia coli in Microchannel by Electric Field and Laser Tweezer, Analytica Chimica Acta, accepted (in printing).

- [123] K. Morishima, F. Arai, T. Fukuda, H. Matsuura, K. Yoshikawa: Bio-Micro-Manipulation System for High Throughput Screening and Micro Surger y in Microchannel, Advanced Robotics, submitted.
- [124] K. Morishima, T. Fukuda, F. Arai, H. Ishihara, H. Matsuura, K.Yoshikawa: Noncontact Transportation of DNA Molecule by Dielectrophoretic Force, IEEE P roceedings of Sixth International Symposium on Micro Machine and Human Scie nce, 145-152, 1995-10.
- [125] K. Morishima, T. Fukuda, F. Arai, H. Matsuura, K. Yoshikawa: Noncontact Transportation of DNA Molecule by Dielectrophoretic Force for Mic ro DNA Flow System, Proceedings 1996 IEEE International Conference on Roboti cs and Automation, 2214-2219, Minneapolis, 1996-4.
- [126] K.Morishima, T.Fukuda, F.Arai, H.Matsuura, K.Yoshikawa: Transportaion of DNA Molecule by Dielectrophoretic Force Utilizing the Confo rmational Transition in the Higher Order Structure of DNA, IEEE Proceedings of Seventh International Symposium on Micro Machine and Human Science, 171-176, 1995-10.
- [127] K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Microflow System and Noncontact Transportation of DNA Molecule in Microflow Channels by Dielectrophoretic Force, 1996 ASME International Mechanical Engi neering Congress and Exposition, MICRO-ELECTRO-MECHANICAL SYS-TEMS, DSC-Vol.5 9, 339-344, 1996-11.
- [128] K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Microflow System and Transportation of DNA Molecule by Dielectrophoretic For ce Utilizing the Conformational Transition in the Higher Order Structure of DNA Molecule, Tenth IEEE International Workshop on MICRO ELECTRO MECHANICAL SYSTEMS, 389-394, 1997-1.
- [129] K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Manipulation of DNA Molecule Utilizing the Conformational Transition in the Higher Order Structure of DNA, Proceedings 1997 IEEE International Conference on Robotics and Automation, 1454-1459, 1997-4.
- [130] K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Manipulation of DNA Molecule Utilizing the Conformational Transition in the Higher Order Structure of DNA in Microchannel, France-Japan Workshop "From N ano. to Macroscale science and technology through" Microsystems (N2M'97), 1997-5.
- [131] K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Transportation of DNA Molecule by Dielectrophoretic Force Utilizing the Conf ormational Transition in the Higher Order Structure of DNA Molecule in Micro flow Channels, Tenth International Symposium on High Performance Capillary E lectrophoresis and Related Microscale Techniques, 1997-7.

- [132] K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Screening of Single Eshcerichia coli in Microchannel by Electric Field and L aser tweezer, International Cyber congress on Analytical BioSciences, 1997-8.
- [133] K. Morishima, F. Arai, T. Fukuda, H. Matsuura, K. Yoshikawa: Screening of a single E. coli in Microchannel by Electric Field and Laser tweezer, IEEE Proceedings of Eighth International Symposium on Micro Mechatronics and Human Science, 155-160, 1997-10.
- [134] K. Morishima, F. Arai, T. Fukuda, H. Matsuura, K. Yoshikawa: Bio-Micromanipulation System for High Throughput Screening in Microchannel, 1998 IEEE International Conference on Robotics and Automation, submitted.
- [135] T. Fukuda, K. Morishima, F. Arai: Noncontact Transportation of DNA Molecule by Dielectrphoretic Force for Micr o DNA Flow System, NEW TECHNOLOGY JAPAN, 15, Vol.23, No.9, 95-12-100-05, Dec ember 1995.
- [136] T. Fukuda, F. Arai, K. Morishima: Manipulation of DNA Molecule by Dielectrphoretic Force, Journal of J. Precise Eng., Vol. 63, No. 7, 963-966, 1997 (in Japanese).

List of Publications

- I Transactions and Journals
 - T. Fukuda, K. Morishima, F. Arai, H. Matsuura, K. Yoshikawa: Noncontact Manipulation of DNA Molecule (1st Report, Transportation of DNA Molecule by Dielectrophoretic Force), Trans. of JSME, Vol. 62, No. 599, 249-256, 1996-7 (in Japanese).
 - 2 K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Transportation of DNA Molecule by Dielectrophoretic Force in Micro Flow Channel -Utilization of Conformational Transition in the Higher Order Structure of DNA-, Trans. IEE of Japan, Vol. 116-E, No. 9, 387-394, Nov., 1996 (in Japanese).
 - 3 K. Morishima, F. Arai, T. Fukuda, H. Matsuura, K. Yoshikawa: Screening of Single Escherichia coli in Microchannel by Electric Field and Laser Tweezer, Analytica Chimica Acta, accepted (in printing).
 - 4 K. Morishima, F. Arai, T. Fukuda, H. Matsuura, K. Yoshikawa: Bio-Micro-Manipulation System for High Throughput Screening and Micro Surger y in Microchannel, Advanced Robotics, submitted.
- **II** International Conference
 - K. Morishima, T. Fukuda, F. Arai, H. Ishihara, H. Matsuura, K. Yoshikawa: Noncontact Transportation of DNA Molecule by Dielectrophoretic Force, IEEE P roceedings of Sixth International Symposium on Micro Machine and Human Scie nce, 145-152, 1995-10.
 - 2 K. Morishima, T. Fukuda, F. Arai, H. Matsuura, K. Yoshikawa: Noncontact Transportation of DNA Molecule by Dielectrophoretic

Force for Mic ro DNA Flow System, Proceedings 1996 IEEE International Conference on Roboti cs and Automation, 2214-2219, Minneapolis, 1996-4.

- 3 K. Morishima, T. Fukuda, F. Arai, H. Matsuura, K. Yoshikawa: Transportaion of DNA Molecule by Dielectrophoretic Force Utilizing the Confo rmational Transition in the Higher Order Structure of DNA, IEEE Proceedings of Seventh International Symposium on Micro Machine and Human Science, 171-176, 1995-10.
- 4 K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Microflow System and Noncontact Transportation of DNA Molecule in Microflow Channels by Dielectrophoretic Force, 1996 ASME International Mechanical Engineering Congress and Exposition, MICRO-ELECTRO-MECHANICAL SYSTEMS, DSC-Vol. 5 9, 339-344, 1996-11.
- 5 K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Microflow System and Transportation of DNA Molecule by Dielectrophoretic For ce Utilizing the Conformational Transition in the Higher Order Structure of DNA Molecule, Tenth IEEE International Workshop on MI-CRO ELECTRO MECHANICAL SYSTEMS, 389-394, 1997-1.
- 6 K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Manipulation of DNA Molecule Utilizing the Conformational Transition in the Higher Order Structure of DNA, Proceedings 1997 IEEE International Conference on Robotics and Automation, 1454-1459, 1997-4.
- 7 K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Manipulation of DNA Molecule Utilizing the Conformational Transition in the Higher Order Structure of DNA in Microchannel, France-Japan Workshop "From N ano. to Macroscale science and technology through" Microsystems (N2M'97), 1997-5.
- 8 K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Transportation of DNA Molecule by Dielectrophoretic Force Utilizing the Conf ormational Transition in the Higher Order Structure of DNA Molecule in Micro flow Channels, Tenth International Symposium on High Per-

formance Capillary E lectrophoresis and Related Microscale Techniques, 1997-7.

- 9 K. Morishima, T. Fukuda, F. Arai, K. Yoshikawa: Screening of Single Eshcerichia coli in Microchannel by Electric Field and L aser tweezer, International Cyber congress on Analytical BioSciences, 1997-8.
- 10 F. Arai, K. Morishima, T. Kasugai, T. Fukuda: Bio Micro Manipulation (New Direction for Operation Improvement), Proc. Int. Conf. on Intelligent Robotics and Systems (IROS), Vol. 3, 1300-1305, 1997-9.
- 11 K. Morishima, F. Arai, T. Fukuda, H. Matsuura, K.Yoshikawa: Screening of a single E. coli in Microchannel by Electric Field and Laser tweezer, IEEE Proceedings of Eighth International Symposium on Micro Mechatronics and Human Science, 155-160, 1997-10.
- 12 K. Morishima, F. Arai, T. Fukuda, H. Matsuura, K. Yoshikawa: Bio-Micromanipulation System for High Throughput Screening in Microchannel, 1998 IEEE International Conference on Robotics and Automation, submitted.

III Review

- T. Fukuda, K. Morishima, F. Arai: Noncontact Transportation of DNA Molecule by Dielectrphoretic Force for Micr o DNA Flow System, NEW TECHNOLOGY JAPAN, Vol.23, No.9, 95-12-100-05, December 1995, 15.
- 2 T. Fukuda, F. Arai, K. Morishima: Manipulation of DNA Molecule by Dielectrphoretic Force, Journal of JSPE, Vol. 63, No. 7, 963-966, 1997 (in Japanese).